Alma Mater Studiorum – Università di Bologna



PhD in Bioengineering (09/G2 - ING-INF/06)

XXV Cycle

TRAINING AND ASSESSMENT OF PATIENTS WITH NEUROLOGICAL IMPAIRMENTS USING VIRTUAL REALITY, AUGMENTED FEEDBACK, INERTIAL SENSING AND LOW COST VIDEO TECHNOLOGY

Eng. Agnese Peruzzi

PhD Coordinator

Prof. Angelo Cappello

Supervisor

Prof. Ugo Della Croce

Co-Supervisor

Prof. Anat Mirelman

Thesis defense: 12th April 2013

Alma Mater Studiorum – Università di Bologna



PhD in Bioengineering (09/G2 - ING-INF/06)

XXV Cycle

TRAINING AND ASSESSMENT OF PATIENTS WITH NEUROLOGICAL IMPAIRMENTS USING VIRTUAL REALITY, AUGMENTED FEEDBACK, INERTIAL SENSING AND LOW COST VIDEO TECHNOLOGY

Eng. Agnese Peruzzi

PhD Coordinator

Supervisor

Co-Supervisor

Internal examiner

External examiner

Prof. Angelo Cappello

Prof. Ugo Della Croce

Prof. Anat Mirelman

aust

Prof. Angelo Cappello

pelolef

Prof. Judith Deutsch

Juditi Dakil 3-17-13

Aí míeí

SUMMARY

Neurorehabilitation is an active process by which individuals affected by neurological impairments achieve a full recovery or, when this is not possible, realize their optimal physical, mental and social potential. The rehabilitation process can be considered successful when it improves daily activities of patients treated and, therefore, their quality of life. The essential components of an effective motor rehabilitation process include an expert multidisciplinary assessment, a realistic and goal-oriented programs and the evaluation of the patient's achievements through clinically appropriate, scientifically sound outcome measures. The main aim of this thesis was to develop quantitative methods and tools for the clinical treatment and assessment of patients with neurological impairments.

Conventional rehabilitation treatments typically require neurological patients to execute repetitive exercises over a long period, with consequences in their engagement and motivation. Virtual reality and augmented feedback are tools recently applied to motor rehabilitation, which have shown to involve patients, allowing repeatability and standardization of protocols. An augmented feedback tool based on the use of IMU for trunk control was developed and the evaluation of its usability is presented in this work.

Moreover, virtual reality allows individualizing the treatment according to the patient needs gradually adapting the difficulty level to the progress. A virtual reality-based application for gait rehabilitation was developed and tested in a 6-weeks training program on patients affected by multiple sclerosis. Usability, feasibility and acceptance of both system and training protocol were evaluated and the effectiveness of the treatment demonstrated.

Traditionally, the quantitative assessment of motor skills has been performed using motion capture systems. Unfortunately, their effective use in clinical practice is still limited by their cost, and the space, time and expertise required operating. Moreover, the application of markers on the body of patients may influence their natural movements. To overcome these limitations the use of inertial sensors has been introduced in the field. A methodology based on their use in assessing the arm swing in subjects affected by Parkinson's disease is proposed and preliminary results show their ability in distinguishing healthy from pathological subjects.

Inertial sensors, which are small, accurate, flexible and portable, have the drawback to accumulate significant drift during long measurements. This particular issue has been analyzed in this work and findings suggest that drift and its consequences in determining gait parameters can be contained if the inertial unit is placed on the foot and accelerations are integrated starting from the mid stance phase of gait.

Finally, a validation of the Microsoft Kinect in tracking gait in a virtual reality-based training is presented. Preliminary results allow defining the range of use of the sensor for applications in rehabilitation.

SOMMARIO

La neuroriabilitazione è un processo attivo attraverso il quale gli individui affetti da patologie neurologiche mirano al conseguimento di un recupero completo o, quando ciò non è possibile, alla realizzazione del loro potenziale ottimale benessere fisico, mentale e sociale. Il processo di riabilitazione può essere considerato effettivo quando migliora le attività quotidiane dei pazienti trattati e, di conseguenza, la loro qualità di vita. Gli elementi essenziali di un processo di riabilitazione motoria efficace comprendono una valutazione clinica da parte di un team multidisciplinare, un programma riabilitativo realistico e orientato a pochi specifici obiettivi e la valutazione dei risultati conseguiti dal paziente attraverso l'intervento mediante misure scientifiche e clinicamente appropriate.

L'obiettivo principale di questa tesi è stato sviluppare metodi quantitativi e strumenti per l'intervento riabilitativo e la valutazione clinica di pazienti affetti da deficit neurologici.

I trattamenti riabilitativi convenzionali tipicamente richiedono a pazienti neurologici l'esecuzione di esercizi ripetitivi per un lungo periodo, andando a incidere negativamente sul loro impegno e sulla loro motivazione. La realtà virtuale e i feedback aumentati sono approcci recentemente adottati dalla riabilitazione motoria, in grado di coinvolgere i pazienti nel trattamento, permettendo la ripetibilità e la standardizzazione dei protocolli. In questo lavoro è stato sviluppato uno strumento basato sull'utilizzo di feedback aumentati e sensori inerziali per il controllo del tronco ed è stata presentata una valutazione sulla sua efficacia e usabilità.

Inoltre, la realtà virtuale permette individualizzare il trattamento in base alle esigenze del paziente, adeguando gradualmente il livello di difficoltà ai suoi progressi. Un'applicazione basata sulla realtà virtuale per la riabilitazione del cammino è stata sviluppata ed è stata poi testata su pazienti affetti da sclerosi multipla durante un programma riabilitativo di sei settimane. Sono state valutate l'usabilità, la fattibilità e l'accettazione sia del sistema sia del protocollo riabilitativo ed è stata dimostrata l'efficacia del trattamento.

Tradizionalmente, la valutazione quantitativa delle capacità motorie dei pazienti viene effettuata utilizzando sistemi di *motion capture*. Purtroppo, il loro uso nella pratica clinica è ancora limitato a causa del loro costo, dello spazio, del tempo e delle competenze necessari per il loro funzionamento. Inoltre, l'applicazione di marcatori sul corpo dei pazienti può influenzare il loro movimento naturale. Per superare queste limitazioni è stato introdotto in questo campo l'utilizzo di sensori inerziali. In questa tesi, viene proposta una metodologia basata sul loro uso nella valutazione dell'oscillazione delle braccia in soggetti affetti da morbo di Parkinson, e si mostra come essa sia in grado di distinguere tra soggetti sani e patologici.

I sensori inerziali, che sono piccoli, accurati, flessibili e portatili, presentano però l'inconveniente di accumulare *drift* rilevanti durante lunghe misurazioni. In questo lavoro è stato affrontato questo particolare problema e i risultati ottenuti suggeriscono che, se l'unità inerziale è posizionata sul piede e le accelerazioni sono integrate iniziando dalla fase di *mid stance* del cammino, il *drift* e le sue conseguenze nella determinazione dei parametri del cammino sono contenuti.

Infine, è stata presentata una validazione del Microsoft Kinect in un'applicazione per il *tracking* del cammino in ambiente virtuale. Risultati preliminari consentono di definire il campo di utilizzo del sensore per applicazioni in riabilitazione.

Table of Contents

SUMMARY IV			
SOMMARIO VI			
1. INTRODUCTION1			
1.1. RATIONALE1			
1.2. OUTLINE			
2. REHABILITATION IN NEUROLOGICAL IMPAIRMENTS			
2.1. NEUROLOGICAL IMPAIRMENTS			
2.2. NEUROLOGICAL REHABILITATION			
2.2.1. The rehabilitation cycle			
2.3. REHABILITATION METHODS			
2.3.1. Motor deficits			
2.3.2. Reduced mobility and gait deficits10			
2.3.3. Neuropsychological deficits			
2.3.4. Dual task during walking			
3. TECHNOLOGICAL TOOLS FOR DESIGNING NEW MOTOR			
REHABILITATION METHODS16			
3.1. MEASURING HUMAN MOVEMENT IN CLINICAL PRACTICE: INERTIAL SENSING			
3.2. VIRTUAL REALITY IN MOTOR REHABILITATION INTERVENTIONS			
3.2.1. What is Virtual Reality			
3.2.2. Virtual Neurorehabilitation			
3.2.3. Motion tracking in virtual environments			
3.2.4. Augmented feedback26			
4. ESTIMATION OF STRIDE LENGTH USING AN IMU: A VALIDATION OF			
THE ZERO VELOCITY ASSUMPTION			

4.1.	INTRODUCTION	
4.2.	Methods	29
4.3.	RESULTS	31
4.4.	DISCUSSION	
5. VI	RTUAL REALITY AND INERTIAL SENSING IN MOTOR TRAININ	G:
APPLI	CATION TO MULTIPLE SCLEROSIS	35
5.1.	INTRODUCTION	35
5.2.	Methods	
5.2	2.1. Experimental set-up	
5.2	2.2. VR environment	
5.2	2.3. Study design and participants	41
5.2	2.4. Rehabilitation intervention	41
5.2	2.5. Outcomes	
5.2	2.6. Statistical analysis	
5.3.	RESULTS	46
5.3	8.1. Evaluation of the VR-based system	47
5.3	<i>B.2.</i> Evaluation of setup and administration of the gait training program	
5.3	3.3. Gait analysis	
5.3	<i>8.4. Endurance, balance and obstacle negotiation</i>	51
5.3	8.5. Neuropsychological tests	
5.4.	DISCUSSION	52
5.5.	CONCLUSIONS	55
6. IN	ERTIAL SENSING IN MOTOR ASSESSMENT: ASSESSMENT OF	
REDUCED ARM SWING IN PARKINSON'S DISEASE		
6.1	I. Introduction	57
6.2	2. Methods	58
6.3	3. Preliminary results	58

6.4. Conclusions	59
7. LOW COST VIDEO TECHNOLOGY FOR TRACKING IN VIRTUAL	REALITY
APPLICATIONS: THE MICROSOFT KINECT	60
7.1. KINECT-BASED GAIT TRACKING FOR VR CLINICAL APPLICATIONS	61
7.1.1. Methods	63
7.1.2. Preliminary results	64
7.1.3. Discussion	64
8. AUGMENTED BIOFEEDBACK: A TOOL FOR TRUNK MOTOR COM	TROL 66
8.1. AUGMENTED FEEDBACK IN NEUROLOGICAL IMPAIRMENTS	66
8.2. INERTIAL-BASED VISUAL FEEDBACK TOOL FOR TRUNK CONTROL	67
8.2.1. Methods	69
8.2.2. Preliminary results	70
9. CONCLUSIONS	72
APPENDIX	
I. ARMSWING GUI	73
II. TRIALS ANALYSIS GUI	75
BIBLIOGRAPHY	76

1. Introduction

1.1. Rationale

The essential components of an effective motor rehabilitation program for subjects affected by neurological impairments include expert multidisciplinary assessment, realistic and goaloriented interventions, and the evaluation of the impact of the rehabilitation through the use of clinically appropriate, scientifically sound outcome measures. Traditionally laboratory based motion capture systems have been used to assess the patient before and after the rehabilitative intervention. This approach usually requires the use of markers to be positioned on the patient's body surface. In some occasions, the presence of markers may represent a source of uneasiness and discomfort, influencing and interfering with the natural movements of the subject. Moreover, motion capture systems, which are generally very expensive, require expertise, space and a long time for set-up, further increasing the cost of the evaluation. All the above reasons help to explain the limited use of these systems in clinical practice, where the evaluations are mostly qualitative.

Rehabilitative interventions of neurological patients typically aim at recovering motor skills. It has been demonstrated that motor recovery is promoted by intensive skillful practice (Shumway-Cook, 2007) and increasing difficulty (Malouin, 2003). Therefore conventional physical therapies in clinical practice are often characterized by the regular and intense repetition of exercises over a long period, consequently limiting patients' engagement and motivation.

Recently, innovative technology and approaches have been proposed to overcome the abovementioned limitations. The main goal of the project described in this thesis was to develop new quantitative rehabilitation methods and tools to be used in the clinical assessment and treatment of patients with neurological deficits.

1.2. Outline

The thesis is organized as follows.

Chapter 2 is an overview of the process of rehabilitation for people affected by neurological impairments.

Chapter 3 describes some of the limitations of conventional motor rehabilitation and proposes new methods and tools to overcome them.

Chapter 4 presents a study on the estimation of the stride length using an inertial measurement unit (IMU). The study evaluates the error associated with the zero velocity assumption in estimating stride length using several locations of the IMU on the foot and on the shank.

Chapter 5 presents a study focusing on a virtual reality (VR)-based system for gait training in subjects affected by multiple sclerosis (MS) using IMUs. The proposed system and training are explained in detail and their feasibility, acceptance and effectiveness are evaluated.

Chapter 6 presents a study on the measurement of the arm swing in Parkinson's disease (PD) patients using IMUs. The proposed methodology has been used both on patients and healthy subjects and preliminary results are presented.

Chapter 7 describes a tracking methodology based on the use of the Microsoft Kinect for the reproduction of gait in a VR application. The proposed methodology is validated using stereo-photogrammetry.

Chapter 8 describes an augmented feedback tool for trunk control based on the use of inertial sensing. Preliminary results on the usability of the tool on young healthy subjects are presented.

Chapter 9 summarizes advantages and limitations of the developed tools and the methods and suggests options for future research.

2

2. Rehabilitation in neurological impairments

2.1. Neurological impairments

The *World Health Organization* (WHO) defines impairment as any loss or abnormality in body structure or of a physiological or psychological function (World Health Organization, 2001). Specifically, all the disorders originating from structural, biochemical or bioelectrical abnormalities in the central and peripheral nervous system lead to neurological impairments. They may cause physical or mental problems, affecting an individual's speech, motor skills, vision, memory, muscle function or learning abilities. Not all neurological impairments are present from birth. A neurological impairment can be acquired as a result of some form of brain or spinal cord injury. Often, the results are very similar; the only difference is the way in which a given part of the brain becomes damaged. Because of its various forms, neurological impairment can be classified in many different ways.

Neurological disorders are diseases of the central and peripheral nervous system, which affect learning and behavioral abilities (World Health Organization, 2007).

Some of the more common neurological diseases (Hirtz, 2007) include multiple sclerosis (MS) and Parkinson's disease (PD), with a crude incidence rate, respectively, of 2.5 (with a range of 1.1–4) and 4.5–19 per 100,000 population per year, and a median estimated prevalence, respectively, of 30 (with a range of 5–80) and 100-200 per 100,000 (World Health Organization, 2007) (World Health Organization, 2008). MS and PD are considered neurodegenerative disorders (Trapp, 1999), (World Health Organization, 2001), (Gao, 2008) and will be particularly considered in the present thesis. Neurodegenerative diseases are defined as conditions characterized by progressive nervous system dysfunction. These disorders are often associated with atrophy of the affected central or peripheral structures of the nervous system.

MS is an inflammatory demyelinating condition of the central nervous system that is generally considered to be autoimmune in nature (World Health Organization, 2007). In

people with MS, the immune trigger is unknown, but the targets are myelinated central nervous system tracts. MS can lead to a wide variety of motor and cognitive symptoms, depending on the affected part of the body and on the severity.

Approximately 80% of individuals affected by MS initially present a form of this pathology called relapsing/remitting (Figure 1), which is characterized by unpredictable attacks, called relapses, during which new symptoms appear or those existing become more severe (Lidcombe, 2003). They can last for long periods (days or months) and are followed by a partial or total recovery, i.e., the remission.



Figure 1 Example of two possible courses of MS in relapsing/remitting form.

The disease may appear clinically inactive for months or years, though a more frequent asymptomatic inflammatory activity is usually present. Over time, however, symptoms may become more severe with less complete recovery of function after each attack.

MS is usually accompanied by physical disability, complicated by fatigue, depression and possibly cognitive impairment and can lead to a functional decline. Typically disease onset is around 30 years of age, hence the loss in functional ability of patients with MS can be substantial and often prevents MS subjects from performing their customary roles.

PD is a chronic progressive neurodegenerative disorder of insidious onset (World Health Organization, 2007), (Gao, 2008). The motor symptoms become readily apparent and diagnosis is made based on cardinal predominate motor signs, however, non-motor features of the disease are increasingly recognized (Chaudhuri, 2006). PD is also associated with late-

onset motor symptoms, such as postural instability and falls, freezing of gait, speech and swallowing difficulties (Jankovic, 2008).

The pathophysiology of PD involves the progressive loss of dopamine-containing neurons of the pars compacta of the substantia nigra that lead to denervation of the nigrostriatal tract and a significant reduction of dopamine at the striatal level. The consequence of this denervation process is an imbalance in the striato-pallidal and pallido-thalamic output pathways, which is responsible for the major motor deficits (Albin, 1989).

Beyond motor and cognitive impairments, quality of life in PD also deteriorated significantly with increasing disease severity particularly in those aspects related to physical and social functioning (Schrag, 2001).

2.2. Neurological rehabilitation

2.2.1. The rehabilitation cycle

The WHO defines rehabilitation as an active process by which those affected by injury or disease achieve a full recovery or, when this is not possible, realize their optimal physical, mental and social potential and are integrated into their most appropriate environment (World Health Organization, 1981). Consequently, a rehabilitation process can be considered successful when it improves independence and quality of life of treated subjects, by maximizing their ability and participation.

The essential components of an effective rehabilitation include expert multidisciplinary assessment, realistic and goal-oriented programs and evaluation of impact on the patient's rehabilitation achievements through the use of clinically appropriate, scientifically sound outcome measures incorporating the patient's and the family's perspectives (European Multiple Sclerosis Platform and Rehabilitation in Multiple Sclerosis, 2004).

The rehabilitation process should be provided by a multidisciplinary team, usually including doctors, nurses, therapists, clinical neuropsychologists and social workers. The aim of the rehabilitation team is to treat impaired body structures to overcome diminished functions, while maximizing patients' activity and participation in their social setting and minimizing

the risk of further symptoms and disability, the patient's pain and the stress of the family and/or caregivers (Stucki, 2002).

The rehabilitation process, seen both from the guidance and from the service perspective, can be thought as an iterative process involving the following elements (Figure 2):

- 1. assessment of the nature of the patient's problems and needs;
- 2. assignment to an intervention program;
- 3. plan and implementation of the assigned intervention;
- 4. evaluation of the intervention.



Figure 2 The rehab-cycle (Stucki, 2002), (World Health Organization, 2007)

ASSESSMENT

The assessment includes the identification of the impairments of patients, which contribute to the difficulties in function, the estimation of rehabilitation potential and prognosis and the definition of the goals of the intervention program. Since neurological subjects suffer from a combination of multiple impairments, it is crucial, to identify target problems, which may be treated and relieved in a reasonable amount of time. In this phase, information is collected through both standardized assessment and interviews.

The most commonly used measures of impairment in neurological patients include: a) the Motricity and the Ambulation Index for motor deficits; b) the Berg Balance test (Berg, 1995) and Timed Up and Go (Podsiadlo, 1991) for balance; c) the Beck Depression Index for depression (Lykouras, 1998); d) the Short Orientation Memory Concentration Test

(Katzman, 1983) and Mini Mental State Examination (Folstein, 1975) for brief cognitive screening e) the Frontal Assessment Battery (Dubois & Slachevsky, 2000), for neurological and cognitive features, and the Montreal Cognitive Assessment (Nasreddine, 2005) a cognitive screening tool. The two most widely used measures of disability are the Barthel Activities of Daily Living index (Collin, 1988) and the Functional Independence Measure (Dodds, 1993).

ASSIGNMENT

The assignment is the determination of a realistic intervention program, identifying and addressing it toward the factors with the greatest potential for improvement. The first aim of the assignment is to make the patient conscious about her/his disability and to stimulate her/him to have an active role on it. The second aim is to define the compensatory strategies for cognitive and behavioral deficits. The third aim consists in planning interventions aiming to generalize the compensatory strategies in the environment. Effective teamwork requires that all team members work towards common goals. Good rehabilitation practice should involve the patient (and the family, when appropriate) in the setting of meaningful, challenging and achievable short-term and long-term goals.

INTERVENTION

The task-oriented approach refers to the specification of the rehabilitative techniques, indicator measures and the target values to be achieved in a predefined time period. Practice of an impaired function usually involves repetition, starting with simple tasks and slowly increasing the level of difficulty (Malouin, 2003). An alternative approach is to use compensatory techniques, to help the patient to achieve the goal by different means. There has been a shift from exercising isolated impairments towards task-oriented therapy for activities of daily living (Horak, 1990).

EVALUATION

The evaluation refers to the assessment of goal achievement with respect to predefined treatment goals. In rehabilitation, the measurement of functioning and health is not only relevant to evaluate intervention outcomes, but the assessment and interventional

management as well. Thus these measures are examined much more closely both at the level of individual problems and at the level of instrumented scales.

Although all the phases of the cycle are present in the choice and implementation of a rehabilitation treatment, in this thesis, the focus will be prevalently on the stages of assessment and rehabilitation intervention, in the effort to make quantitative the outcome coming from them.

2.3. Rehabilitation methods

Rehabilitation should start as soon as possible after the diagnosis of a neurological impairment and should focus on the community rehabilitation perspective (World Health Organization, 2007). Neurological patients can present a wide number of complexities including physical functioning limitations, cognitive and communication impairments, behavioral problems, compromised basic daily living activities and psychosocial limitations. Consequently, intervention programs and services have been developed, showing to contribute effectively to the optimal functioning of people with neurological conditions. Here we will briefly discuss those aiming to treat motor and neuropsychological deficits.

2.3.1. Motor deficits

The motor activity in neurological diseases acquires importance for the neuro-motor features of the movement that, although impaired, can be used to recondition and reach a new balance on coordinative pre-existing movement patterns. The physical exercise has shown to provide beneficial effects in neurodegenerative diseases like PD, (Bergen, 2002) but also in MS, where endurance exercises have been showed to make higher the threshold of the fatigue perceived by the patients (Gutierrez, 2005), contributing to improve their quality of life (Petajan, 1996), (Brown, 2005).

Rehabilitation in neurodegenerative disorders stands as its primary objective the improvement of the neuro-muscle-skeletal diseases, but also the enhancement of the global motor performance, with the consequent reduction of fall risk (Valobra, 2000). Therapeutic

approaches include stretching and strength training, rehabilitation of postural transitions, reeducation of postural instability and gait, and training to cognitive strategies.

One of the most common motor dysfunction encountered in the motor rehab of MS is muscle weakness, with onset in 90-100% of cases (Valobra, 2000). It is therefore important to maintain the level of physical performance for as long as possible, by a process that does not generate fatigue or changes in body temperature.

Postural instability is also a disorder consistently reported by MS patients, with onset in 23-82% of cases, (Valobra, 2000), and can be addressed with specific or non-specific rehabilitative approaches, which are based on intensive, repetitive tasks and augmented feedback. Specific treatments aim to learn patients of motor strategies pointing to improve their residual potential in postural control. Non-specific treatments include generic mobilization and stretching exercises, in order to maintain range and muscle elasticity, especially in the lower limbs.

Moreover, about the 55% of individuals with MS identify fatigue as one of the major symptoms affecting their mobility (Fisk, 1994). The exact origin of fatigue is unclear, although it could be sought in the association of weakness, spasticity, ataxia, depression, and heat (Jacobs, 1986).

Other common motor symptoms in MS are spasticity, paresis, stiffness, sensory impairments and fasciculation (Zuvich, 2009). Spasticity (onset 75% of cases) can become a problem as it often interferes with rehabilitation. After that the sources of nociceptive input have been identified and treated, the patient should be instructed to perform appropriate muscle stretching exercises aiming at expand the range of motion of the affected joints (Merritt, 1981). After, the facilitation of motor strength can be accomplished by using passive and active therapies, advancing to progressive resistance training. In active therapy the patient is encouraged to perform an exercise, while the therapist provide her/him a feedback about her/his performance, in order to avoid unnecessary muscle activity and to achieve selective control of specific muscles. Passive therapies include positioning patients on their side, sitting, or standing in a support frame, and stretching exercises.

Recurrent motor symptoms in PD, on which the rehabilitation has shown to be mostly effective, are bradykinesia, rest tremor, rigidity, and postural disturbances and, in late-onset, postural instability and falls and freezing of gait (Jankovic, 2008). The hypertonic muscles and bradykinesia aggravate the condition of the muscle-skeletal system, already altered by age (Valobra, 2000). Muscle stretching is definitely a basic treatment also in PD and the districts where it is prevalently needed are the flexor muscle groups of trunk and limbs. Stretching has the effect of lengthening the slow and progressive soft tissues and in presence of severe flexion contractures, where traditional stretching is not effective, prolonged statics stretching sessions could be required. This technique consists in maintaining tolerable postures for long periods, in order to act more markedly on shortened connective tissues. Joint mobilization is an imperative treatment modality for a disease that has immobility as the main effect. This method can be performed in passive mode, active-assisted or preferably active, depending on the degree of autonomy of the patient in the execution of exercises.

2.3.2. Reduced mobility and gait deficits

Neurological disorders, due to the characteristics of the disease, such as type, location, extension, stability or progressivity and age of onset, are characterized by an extreme variety of clinical manifestation, resulting in alterations of the motor control which cause, during walking, different changes in the kinematics of the upper and lower limbs and trunk. Regardless of the pathology, the ability of gait is modified in terms of:

- effectiveness: reduction in the rate of spontaneous and maximum sustainable speed;
- security: the need for assistance and / or supervision and / or direct assistance;
- efficiency: increase of energy cost;

with consequent limitations of autonomy and participation in social and working life. The observational analysis and the global clinical segmental assessment allow defining the main problems of gait in terms of global imbalances, focal problems and coordination. Reduced mobility is one of the most frequent impairment in MS patients and may be due to motor, sensory, balance and exploratory deficits. The aim of the rehabilitation project may be restoration or maintenance of the residual skills, using also orthoses or assistive devices. Trunk stability represents a fundamental prerequisite for locomotor rehabilitation; therefore a preparation program to standing, performing exercises on the mat, usually precedes any gait treatment. Most of these activities aim at the reduction of the base support, the verticalization, the maintenance of postural stability, in both static and dynamic conditions. Once that a good alignment trunk-limbs is reached in the standing position, dynamic stability will be trained and obtained with destabilization-stabilization movements of the trunk. As soon as a suitable motor control in the upright position has been reached, it is possible to start to walk handling the bars, trying to reset the motor patterns of gait, coordinating the movement of the lower limbs with the pendular synkinesis of the arms.

Gait reduction plays an important role also in the rehabilitation of PD subjects (Valobra, 2000). The patient is trained to use the most correct gait pattern as possible, which is the more stable and the less energetically expensive. In PD, particular attention should be paid on direction changes, often usually carried out pivoting on a limb and causing instability. The patient should be instructed to change direction, particularly when turning back, touring enough largely to prevent the two legs stepping on each other. The path must be trained not only on smooth terrain such as gyms, but also on uneven surfaces. To make the rehabilitation treatment as effective as possible in real life, it is important to include also exercises upward and downward by a step, and training aiming at negotiating obstacles of different height and shape.

In treadmill (TM) training with partial body-weight support (BWS) the patient is secured into a harness hanging above a TM on which she/he walks. The use of TM finds in PD a beneficial application because it forces the subject to a rhythmic path. Indeed, this intervention has shown to improve gait in individuals affected by neurodegenerative disorders, such as PD (Mehrholz, 2010), (Miyai, 2002) and MS (Benedetti, 2009), (Pilutti, 2011)

For a further diagnostic-functional inspection, clinical examination can be extended to instrumental analysis, which is addressed to quantify:

- The dynamic neuromuscular pattern using dynamic electromyography;
- The global and segmental kinematic characteristics using optoelectronic systems;
- The global and segmental kinetic through force platforms and optoelectronic systems;
- The energy cost of ambulation through systems of measurement of oxygen consumption.

The three-dimensional optoelectronic analysis quantifies the spatial-temporal gait parameters, kinematics and kinetics segmental. The three main categories of problems can be scaled according to the deviation from normality data and are (Bowden, 2006):

- Reduced load acceptance;
- Reduced propulsion;
- Reduced load acceptance and reduced propulsion.

2.3.3. Neuropsychological deficits

Most neuropsychological interventions are related to the treatment or optimization of cognitive deficiencies including also emotional, behavioral and personality alterations, aiming at the best cognitive, neurobiological and social re-adaptation (Valobra, 2000).

Problems in memory and concentration are very common reported symptoms in neurodegenerative disorders and affect tasks such as decision-making, planning, sensory integration and dual tasking (Yogev-Seligmann, 2008). The memory is now considered as a function consisting of several components that work sequentially and integrally. We can distinguish between a short-term memory, perspective memory, which relates to knowledge skills that are implicitly acquired, and a long-term memory (Valobra, 2000). The rehabilitation intervention should take into consideration the specificity of patients and the level of their deficit, distinguishing subjects able to learn and apply new compensation strategies, from subjects able to acquire only implicit behaviors. According to literature of recent years, about half of MS patients are affected by memory disorders, as well as by attention, executive and visual-spatial functions (Rao, 1991). Disorders in long- and short-

term memory, saving the implicit memory, characterize these patients and are, with attention functions and speed processing, the first symptoms of the disease. Specifically, like in PD, the main cognitive affected areas include memory, executive functions and attention (Chiaravalloti, 2008), (Caballol, 2007).

Attention constitutes a phenomenon with more components including selectivity and intensity. The first dimension includes two separate processes: the focused attention, generally tested by asking to the subject to find a specific target stimulus among several distractors, and divided attention, assessed by dual task, where the individual must respond to two stimuli interfering with each other. The intensity does include the warning and physical alertness. The alert aims to implement a response following a warning signal, while the supervisory indicates the ability to maintain the appropriate response for a certain period of time. Batteries are standardized to highlight any shortfall depending on attention processes. The rehabilitation protocols can be gathered in two groups: the first one aims to recover the lost capacity, while the second aims to develop compensatory strategies.

Also neuropsychiatric and behavioral disorders are common in neurodegenerative disorders. Depression, for example, seriously affects the autonomy and quality of both individuals affected by MS (Arnett, 2008) and PD (Mayeux, 1984), (Taylor, 1986).

2.3.4. Dual task during walking

Performing two tasks simultaneously (dual tasking) is a frequent activity in everyday life, which requires divided attention. Indeed, when people attempt dual task, performance is generally impaired, characterized by more errors or longer reaction times than the same task performed without a concurrent task (Wu, 2008). The capacity to execute a second task (dual task performance) is highly advantageous during walking because it allows for communicating with people, transporting objects and monitoring the environment.

The effects of dual tasks on gait (dual task cost) have been studied in several populations, from healthy young and older adults to neurologic patients. In healthy adults, dual tasking has demonstrated to reduce the performance of the concurrent task and to decrease the gait

speed, highlighting how deficits in attention and executive functions are associated with fall risk, postural instability and impairments in daily life activities (Yogev-Seligmann, 2008). The dual task cost has been studied also in elderly people (Woollacott, 2002), showing an increase in the reaction times of the concurrent cognitive task or a decrease in gait speed (Chen, 1996), (Ebersbach, 1995), and an increased gait variability (Dubost, 2006). Agerelated gait changes are more pronounced in people with cognitive impairments (Hausdorff, 1997), (Holtzer, 2006), (Lindenberger, 2000) and are accentuated under dual task condition. Lundin-Olsson et al. showed that the instability to maintain a conversation during walking constitutes a marker of future falls in older adults (Lundin-Olsson, 1997). Several studies have also demonstrated that dual tasking severely affected gait parameters in populations prone to falls, much more then in a healthy elderly people (Bloem, 2001), (Beauchet, 2005), (Toulotte, 2006), (Springer, 2006), (Verghese, 2002). Faulkner et al. observed that changes in performance while dual-tasking could be used to identify subjects at risk for recurrent falls (Faulkner, 2007). A recent study on a large sample of older adults found that the dual task cost of 18% or more predicted falls in individuals walking faster than a specific gait speed threshold (Yamada, 2011).

Typically, the dual task cost is larger in neurological patients than healthy age-matched controls (Sheridan, 2003). This has been largely investigated in PD patients and can be explained by their impairment in attention and in executive function processes (Dubois, 1996) and by their altered gait, typically characterized by slow gait speed, short strides, high double support time (Yogev, 2005) (Morris, 1994), (O'Shea, 2002), (Bond, 2000), (Camicioli, 1998), (Hausdorff, 2003), (Lewis, 2011), decreased symmetry and coordination between left and right steps (Plotnik, 2008), (Yogev, 2007), and increased stride-to-stride variability (Plotnik, 2011), (Yogev, 2005), (Hausdorff, 2003).

Contrarily to PD, the dual task cost during gait in MS has been poorly explored. In a recent case-control study (Hamilton, 2009) on MS subjects, walking and attention interaction have been investigated. Compared to healthy controls, MS subjects demonstrated slower speed and

14

elevated swing time variability in gait performance under cognitive dual task conditions. The authors suggested that fatigue and general cognitive ability contribute to this.

These results seem to be consistent with the differential dual-task decrements reported in another study on MS patients (D'Esposito, 1996), when compared to controls, during the execution of two cognitive tasks simultaneously. Different interpretations can explain these findings, including reduced working memory capacity, task demand, use of different strategies, confounding factors and a divided attention deficit. Moreover, another study (Sosnoff, 2011) investigated the effect of a cognitive task on gait performance in MS individuals with mild, moderate, and severe disability. The findings showed that, compared to patients with mild and moderate disability, the group with severe disability walked slower, with shorter steps, and spent a greater percentage of the gait cycle in double support.

The growing evidence that instability and falls increase during the performance of multiple tasks suggests the need of training balance and gait in dual task in neurological patients. Dual-task training involves the execution of the primary task (maintaining postural control or walking speed) while performing a secondary task, for example a cognitive challenge such as counting backwards, or a manual task such as carrying an item (Woollacott, 2002). Few studies have tried to specifically address this issue, and the findings have not always been consistent (You, 2009), (Yang, 2007), (Canning, 2008), (Silsupadol, 2009), (Schwenk, 2010), (Yogev-Seligmann, 2012), (Plummer-D'Amato, 2012). Still, promising results suggest that training improves dual task gait in older adults with balance impairment (Silsupadol, 2009), patients with dementia (Schwenk, 2010), post-stroke individuals (Yang, 2007), (Plummer-D'Amato, 2012), (Canning, 2008), suggesting that even among patients with neurodegenerative disease, intensive and repetitive practicing of DT while walking can lower dual-task costs.

3. Technological tools for designing new motor rehabilitation methods

The aim of the bioengineering research in motor rehabilitation is to develop new methods for the assessment of impaired patients and, simultaneously, on the basis of the responses gathered from them, to provide clinicians with tools to tailor or facilitate motor rehabilitative interventions (Bonato, 2010). These new approaches must be made with a view to overcoming current limitations in routine clinical practice, where qualitative scales and functional tests are still largely preferred, due to practical reasons, and where treatments are prevalently based on repetition. In this sense, the relevance recently gained by pervasive solutions and personalized interventions for healthcare among researchers and clinicians had a high impact on rehabilitation. New approaches have been primarily possible thanks to the recent progresses in several fields, such as telecommunications, electronics, computer science and real-time data analysis. In particular, a wide range of new technologies, including inertial sensors and low cost video technology, and tools, such as multi-sensory interfaces and virtual reality (VR) have been recently experimented, and in some case combined, opening new perspectives for relevant applications in motor rehabilitation. In this chapter we will introduce them, presenting their potential application in motor rehabilitation of neurological subject.

3.1. Measuring human movement in clinical practice: inertial sensing

The ability to measure human movement quantitatively represents an essential part of clinical assessment and evaluation, thus either allowing for a more complete diagnosis or determining the efficacy of motor rehabilitation interventions.

Motion capture systems, using instruments such as optical motion capture and force plates, are considered as the gold standard in the field of motion analysis for assessing joint kinematics and kinetics. For kinematics, optical motion capture system consists in tracking the position of markers attached to specific locations on the subject's body using a set of cameras, and reconstructing their 3D position. Technology is either based on active or passive markers and uses the red and infrared light range. For kinetics measurements, the reference system used in laboratory setting is force platforms. They measure the ground reaction forces generated by a body standing on or moving across them. They can provide also accurate temporal parameters such as foot initial and terminal contact.

Measuring body movements in laboratory setting under controlled conditions allows getting precise, accurate and reliable measurements of the movement pattern of the subject, and add quantitative and objective figures to the clinical gait assessment. Nevertheless, motion capture systems present also several disadvantages, such as the costly equipment and the need of technical expertise to operate. Another drawback is represented by the confinement of such a system inside the laboratory setting, where the volume of measurement is limited. This aspect can strongly influence the natural behavior of the subjects and does not allow observing them in their everyday life. More importantly, the interpretation of the outcomes is not straightforward and requires further processing and analysis by clinical expertise. Consequently, the use of these devices in the routine clinical practice is limited and it is mostly used for research purpose. In order to overcome them, in the last two decades, new methods (e.g. markerless techniques (Deutscher, 2000) have been successfully implemented. Wearable technology overcomes the limitations of settings and cost, offering an inexpensive, and efficient manner of performing motion analysis in several health-related applications, outside the laboratory. Inertial wearable sensing, (Teng, 2008), (Bonato, 2010) may be used in in the motion analysis of neurological individuals (Bonato, 2009). These sensors, based on a technique for measuring the motion of an object without the need of an external reference, are named Inertial Measurement Units (IMUs) and are composed by different inertial sensor technologies, including accelerometers, gyroscopes and magnetometers. Gyroscopes provide a measurement of the angular velocity applied to the object and thus an estimation of the rotated angle and actual orientation if an initial reference is provided. Though they are usually based on the concept of measuring the Coriolis force, gyroscopes based on other operating principles also exist (electronic, microchip-packaged MEMS gyroscope devices,

solid-state ring lasers, fiber optic gyroscopes, and the extremely sensitive quantum gyroscope). They can be used for the measurement of the motion and posture of any human body segment (Ayrulu-Erdem & Barshan, 2011), (Catalfamo, Ghoussayni, & Ewins, 2010), (Tuncel, Altun, & Barshan, 2009). Since gyroscopes have different sources of dynamic drift, the estimation of the orientation deteriorates with time. To correct these effects, accelerometers and magnetometers are added to the system through data fusion algorithms so that external references are provided for drift correction.

Accelerometers are inertial sensors measuring the linear acceleration along their sensitive axis. Their common operation principle is based on a mechanical sensing element consisting of a proof mass attached to a mechanical suspension system. According to Newton's second Law, under the influence of external accelerations the proof mass deflects from its neutral position and, using the physical changes in the displacement of the proof mass, the acceleration can be measured electrically. Three common types of accelerometers are available, namely, piezoelectric, piezoresistive, and differential capacitive accelerometers (Öberg, 2004), (Mathie, 2004).

Magnetometers are usually based on the magnetoresistive effect. If a magnetic field is applied, a Lorentz force proportional to it will deflect the current path, increasing the resistance. Since the resistance change is proportional to the tilt angle in relation to the magnetic field direction (Graham, 2004), magnetoresistive sensors can estimate changes in the orientation of a body segment in relation to the magnetic North or the vertical axis in the gait analysis (Dai, 1996), (O'Donovan, 2007), (Choi, 2008), providing information that cannot be determined by accelerometers or the integration of gyroscope signals. Accelerometers give a measure of the direction of the gravity vector, and magnetometers provide measurements of the direction of the Earth's magnetic field. With this technology, IMUs are able to accurately estimate their own orientation with respect to a fixed reference frame formed by gravity and the Earth's magnetic North vectors.

IMUs have great potential for measurement of human movement in rehabilitation. They can be used for kinematic measurements (Mayagoitia, 2002), (Luinge, 2005) in ambulatory circumstances. Mathie et al. (Mathie, 2004) reviewed the accelerometer-based systems applied to human movement, distinguishing between monitoring of specific movements (e.g. gait, fall, sit-to-stand transfer etc.), assessment of physical activity and classification of movements. Godfrey et al. (Godfrey, 2008) compared all the techniques that have been used using accelerometers since the early 1990s to 2006 for human movement analysis. To obtain more information on human kinematics, gyroscopes and magnetoresistive sensors have been combined with accelerometers. Gyroscopes are usually used to measure the angular rate and the joints' range of motion (Tong, 1999), (Coley, 2005), (Miyazaki, 1997), while the magnetoresistive sensors provide an additional reference measure for body orientation. Recently, Altun et al. (Altun, 2010) compared the different techniques of classifying human activities using wearable inertial and magnetic sensors.

Following Mathie's classification, IMUs can be used in rehabilitation to:

- a) Monitor specific movements, such gait (Tong, 1999), (Menz, 2003), (Sabatini, 2005), (Tao, 2012), sit-to-stand transfer (Najafi, 2002), falls (Williams, 1998), (Doughty, 2000), (Bourke, 2007), (Wu, 2008);
- b) Measure and assess human motion, for clinical assessment (Wade, 2010), (Parnandi, 2010), (Mancini, 2010), (Palmerini), (Mancini, 2012), for tracking purposes (Lee, 2003), (Zhu, 2004), (Foxlin, 2005), (Zhou, 2008), (Guo, 2009), (Hussain, 2012), and for treatment evaluation (Lorincz), (Jovanov, 2005);
- c) Detect and classify activities (Aminian, 1999), (Najafi, 2003), (Bao, 2004), (Ravi, 2005), (Parkka, 2006), (Preece, 2009), (Yang, 2010).

All these purposes are even more important for individuals affected by neurodegenerative disorders. An increasing interest toward inertial sensing has been recognized in measuring mobility or walking impairment in neurological populations (Pearson, 2004) including persons with MS (Weikert, 2010), (Snook, 2009). For example, quantitative measures coming from inertial units may assist clinicians in assessing gait (Salarian, 2004) and evaluating the above-mentioned turning difficulties of PD subjects, especially in home-based assessments (Boonstra, 2008), (Salarian, 2007). Zampieri et al. has showed that the Timed

Up and Go test, extensively used to assess balance and mobility in moderate-to-severe stage PD, measured with the sole stopwatch was not sensitive enough to detect abnormalities in early-to-mid stage PD and used inertial sensors to make the test more reliable (Zampieri, 2010).

In Chapter 0 an application based on the use of IMUs for motor assessment of PD patients is presented.

3.2. Virtual Reality in motor rehabilitation interventions

One of the main goals of motor rehabilitation is to increase the quantity and quality of patients' daily activities to improve their independent living. Essential for motor recovery is a task-oriented treatment characterized by intense skillful practice (Shumway-Cook, 2007) and increasing difficulty (Malouin, 2003). Consequently, physical therapies usually request patients to regularly execute movement patterns repetitively over an extended period, often limiting their engagement and motivation in rehabilitation. "Motivation is an important factor in rehabilitation and is frequently used as a determinant of rehabilitation out-come. In particular, active engagement towards a treatment/training intervention is usually equated with motivation, and passivity with lack of motivation" (Colombo, 2007). VR is a relatively recent approach in rehabilitation, which demands focus and attention, increasing patients' motivation, and provides them with a sense of achievement (Lange, 2011).

3.2.1. What is Virtual Reality

Coates in 1992 defined VR as "...electronic simulations of environments (...) enabling the end user to interact in realistic three-dimensional situations." (Coates, 1992). To date, VR refers to the use of interactive simulations created with computer hardware and software to present users with opportunities to engage themselves in environments reproducing the real world, (Weiss, 1998). Users interact with the environments, performing actions inside them and/or moving and manipulating virtual objects, in a way that attempts to "immerse" them within the simulation. Immersion is an important concept and relates to the extent to which the VR system succeeds in delivering an environment, which refocuses the user's sensations

from the real to a virtual world (Slater, 1999). A second key concept of VR is the sense of presence, which characterizes the user's interaction within the VR environment. Whereas immersion is an objective measure referring to the VR platform, it does not immediately correspond to the level of presence (which is subjective). VR environments may be delivered to the user via several technologies that differ in the extent to which they are able to "immerse". There is considerable evidence indicating that a high sense of presence may lead to deeper emotional response (Weiss, 2005), increased motivation and, in some cases, enhanced performance (Schuemie, 2001). To provide to the users augmented sensorial feedback (discussed in paragraph 3.2.4) about their performance, helps to achieve a stronger feeling of presence in the virtual world.

VR environments are usually experienced with the aid of special hardware and software for input (transfer of information from the user to the system) and output (transfer of information from the system to the user). The selection of appropriate hardware is important since may greatly influence the way users respond to a VR environment (Rand, 2005). The output to the user can be delivered by different modalities including visual, auditory, haptic, vestibular and olfactory stimuli. Visual information is commonly displayed by Head Mounted Displays (HMDs¹), projection systems or flat screens. Sophisticated VR systems employ more than specialized visual displays, such as audio and haptic² display, engaging the user in the VR environment. Other, less frequently used ways of making the virtual environment more life-like are by letting the user stand on a platform capable of perturbations and thereby providing vestibular stimuli such as the multisensory system CAREN (Motek, Amsterdam). Even less frequent is the provision of olfactory feedback to add odor to a virtual environment (Weiss, 2005).

¹ An HMD is composed of two small screens positioned at eye level within special goggles or a helmet. Thus users view the virtual environment in very close proximity. Advanced HMDs even provide stereoscopic 3D displays of the environment and usually are referred to as more immersive systems (Weiss P. K., 2005).

² Haptic feedback enables users to experience the sensation of touch, making the systems more immersive and closer to the real world experience (Weiss P. K., 2005).

Equally important to achieving a realistic experience within a virtual environment is the ability of the user to navigate and manipulate objects within it. Thus the user must be able to interact (directly or indirectly) with the environment via input technologies. One class of input technologies may be considered as direct methods since users behave in a natural way, and the system tracks their actions and responds accordingly. Generally, tracking (discussed in paragraph 3.2.3) is achieved by using special sensors (non-visual based systems) or by visual tracking (visual based systems). A second class consists of indirect ways for users to manipulate and navigate within a virtual environment. These include activation of computer keyboard keys, a mouse or a joystick or even virtual buttons appearing as part of the environment (Rand, 2005).

Beyond specialized hardware, application software is also necessary. In recent years, "offthe-shelf", "ready-for-clinical-use" VR software has become commercially available. However, more frequently, special software development tools are required in order to design and code an interactive simulated environment that will achieve a desired rehabilitation goal. In many cases, innovative intervention ideas may entail customized programming to construct a virtual environment from scratch, using traditional programming languages (Weiss, 2005).

VR hardware, combined with virtual environments, provides engineers tools for designing rehabilitation interventions customized on patients. When creating a specific VR application for rehabilitation the clinician and technical team face the challenge of choosing and integrating the software and hardware. These decisions are made taking into account budget, physical space, mobility of the system, patient population, complexity of the task with respect to the patient population and the extent of immersion desired from the system. In Chapter 5 a VR application for motor rehabilitation is described in detail.

3.2.2. Virtual Neurorehabilitation

Virtual rehabilitation is the use of VR within rehabilitation (Burdea, 2002). It allows creating environments for assessing and rehabilitating patients, where controlled presentations of

stimuli, motion tracking and performance recording are possible (Rizzo & Kim, 2005).

The rationale for using VR in rehabilitation is based on a number of unique attributes of this technology (Riva, 1999), (Schultheis, 2001). These include the opportunity for active learning, which engages the participant in two or more cognitive and motor activities simultaneously (i.e., dual tasking). This involves and motivates the participant (Mantovani, 2003) and enhances motor learning through problem solving and decision-making. It has been shown that people with impairments seem able to both learn motor skills in VR and transfer the learnt abilities to their real life (Holden, 2005).

In addition, VR allows to objectively quantifying changes and measuring behavior in challenging but safe environments, while maintaining strict experimental control over stimulus delivery and measurement (Rizzo, 2002). VR also offers the possibility to individualize treatments and to standardize assessment and training protocols (Weiss, 2005). VR environments provide the opportunity for repeated learning trials and offer the capability of gradually increasing the complexity of tasks while decreasing the support and feedback provided by the therapist (Schultheis, 2001). Moreover, VR offers the opportunity to provide multi-sensory feedback simultaneously in order to broaden users' rehabilitation. The assumption is that, by displaying and augmenting the same information to different senses, it is possible to increase the amount of knowledge available to participants and consequently assist their performance. Finally, the automated nature of feedback delivery during a VR treatment enables a therapist to focus on the provision of maximum physical support when needed without reducing the complexity of the task.

Among the disadvantages, a factor that may limit the use of VR for patients is cybersickness, which refers to side effects experienced by some users during and following exposure to VR environments (Kennedy, 1997), (Kennedy, 1996). Effects noted while using some VR systems can include nausea, eye-strain and other ocular disturbances, postural instability, headaches and drowsiness. Effects noted up to 12 hours after using VR include disorientation, flashbacks and disturbances in hand–eye coordination and balance (Kennedy, 1997), (Stanney, 1998). Many of these effects appear to be caused by incongruities between

information received from different sensory modalities (Lewis C. a., 1998). Other factors that may influence the occurrence and severity of side effects include characteristics of the user and the display, the user's ability to control simulated motions and interactivity with the task via movement of the head, trunk or whole body (Lewis C. a., 1998).

VR has successfully been used in assessment and rehabilitation of in the neuropsychological and motor deficits in people affected by neurological impairments. Specifically, several studies have been used VR application aiming to train balance and posture, gait, upper and lower extremity function (Sveistrup, 2003). Bisson et al. measured attention demands and functional balance scores before and after a VR balance training programs with augmented biofeedback, demonstrating an improvement of functional balance in traumatic brain injury survivors (Bisson, 2007). VR-based motor training for lower limb has shown encouraging results in post-stroke populations (Laver, 2011), such as improvements in gait speed (Fung, 2006), (Yang, 2008), (Walker, 2010) and stride length (Jaffe DL, 2004), (Mirelman A., 2009), and in PD individuals (Mirelman A. M., 2011). For what concerns the virtual rehabilitation of the upper limb, the majority of the literature is addressed to post-stroke populations and has shown encouraging results (Levin, 2009) (Henderson, 2007), (Holden, 2005).

3.2.3. Motion tracking in virtual environments

In order to involve patients in the virtual motor rehabilitation treatment, it is necessary to immerse them into the simulation, reproducing their limbs movement in real-time inside the virtual environment. A good tracking of the head also contributes to enhance the sense of presence in the simulation, recording the head orientation and moving the virtual environment accordingly (Baillot, 2001).

Features that are usually required to a good real-time tracking are the high accuracy, low encumbrance, high robustness, low invasiveness (users should be unrestricted in their mobility) and minimum latency (Ribo, 2001).

The existing human limb tracking systems can be classified as non-vision based and visionbased systems. Non-vision based systems use inertial, mechanical and magnetic sensors to collect movement signals. Magnetic tracking systems do not require much space but they tend to have limited accuracy due to magnetic field distortions caused by large metal objects common to industrial environments. Inertial measurement systems can be used in many circumstances without limitations (i.e. illumination, temperature, or space, etc.) and show better performance in accuracy against mechanical sensors. With this sensor approach, such as used by Intersense (InterSense Inc., Billerica, MA, USA), InterTrax2, a three degrees of freedom, inertial orientation tracker used to track pitch, roll and yaw movements, the user wears a tracking device that transmits position and orientation data to the VR system. The main drawback of using inertial sensors is that accumulating errors (or drift) can become significant after a short period of time. This issue is discussed in detail in Chapter 4, while a tracking methodology for VR based on the use of IMUs is described in Chapter 5.

In vision-based systems, the user's motion is recorded by video cameras, where special software processes the video image, extracts the user's figure from the background in realtime, and identifies any motion of the body. Unfortunately this approach to human motion tracking often involves intensive computations, such as temporal differencing, background subtraction or occlusion handling (Sen, Leo, Tan, & and Tham, 2011).

The Microsoft Kinect for Windows is a low cost sensor belonging to the vision-based class since it exploits an infrared (IR) structured light to calculate the distances between the IR camera and points in the environment. It consists of three main components, such as IR laser emitter, an IR camera, constituting the depth sensor, and an RGB camera. The inventors describe the measurement of depth as a triangulation process (Freedman, 2010). The Kinect sensor captures depth and color images simultaneously at a frame rate of up to 30 fps. In Chapter 7 an application using the Microsoft Kinect for tracking gait and reproducing it in a VR environment is illustrated.

Remarkable improvements have been achieved by using hybrid tracking systems, which combine the strengths, eliminating the disadvantages, of complementary sensing systems

(e.g. optical and inertial tracking (You, 1999), optical and magnetic tracking (Auer, 1999) (Baltadjieva, 2006) (Behrman, 1998).

3.2.4. Augmented feedback

Reliable sensory information and correct integration of sensory information are necessary for motor control. In neurological impairments, and also as a natural consequence of ageing, this information may be inadequate and, consequently, the control results impaired (Dozza, 2007).

"Biofeedback can be defined as the use of instrumentation to make covert physiological processes more overt" (Huang, 2006). A sensory feedback, augmenting or substituting the sensory movement information, gives to neurodegenerative patients the opportunity to observe a physiological function otherwise not perceptible and regain the ability to better assess different physiological responses and possibly to learn self-control of those responses (Hilgard, 1975).

During the learning process of a motor skill, feedback is the positive or negative response that can inform the learner how well she/he performed the task. The term feedback can be divided in two classes: the inherent (or intrinsic) feedback and augmented (extrinsic) feedback. Inherent feedback is the sensory information that tells the learner how well the task was completed: a basketball player will understand that he/she made a mistake when the ball misses the hoop. In contrast, augmented feedback is information that supplements or "augments" the inherent feedback: for example, when a person is driving over the speed limit and a beep sound is generated by the car. Although the car did not do any harm, the beep gives augmented feedback to the driver in to increase safety (Schmidt, 2005).

The augmented feedback can assume a high number of independent dimensions: it can be concurrent or terminal, immediate or delayed, accumulated or distinct, verbal or non-verbal. An important category of the augmented feedback is the knowledge of results (KR), which is defined terminal augmented feedback about the goal achieved and not about the movement itself. In experimental studies, KR usually refers to a score or, in any case, to information
provided over and above those sources of feedback that are naturally received when a response is made (Adams, Response feedback and learning, 1968).

Knowledge of performance (KP) refers, instead, to information about the quality or patterning of a movement. It may include information such as displacement, velocity or joint motion. KP tends to be distinct from intrinsic feedback and more useful in real-world tasks. It is a strategy often employed by rehabilitation practitioners (Winstein, 1991).

In the next chapters several applications of augmented feedback in virtual environments are described. In particular, in Chapter 8, a visual feedback tool for trunk control is presented.

4. Estimation of stride length using an IMU: a validation of the zero velocity assumption

This chapter was written on the basis of the published article "Estimation of stride length in level walking using an IMU attached to the foot: A validation of the zero velocity assumption during stance" (Peruzzi, Della Croce, Cereatti; Journal of Biomechanics 2011).

4.1. Introduction

Wearable IMUs, including accelerometers and gyroscopes, allow measuring and tracking human locomotion (Sabatini, 2005); (Yun X., 2007), along with the estimation of spatial parameters (such as the stride length), both outdoor indoor and in non-controlled environment and for prolonged periods of time. Once estimated the IMU orientation in the global reference frame (Sabatini, 2005); (Schepers, 2007), linear displacements can be obtained by double integrating the IMU linear coordinate acceleration in the global reference frame and by removing the gravitational contribution from the accelerometer signals. However, the described procedure is complicated by the next factors: (a) a drift commonly present when integrating the accelerometer and gyroscope signals introducing an error in the displacement estimations, which is nonlinearly related to the integration time (Djuric, 2000); (Thong, 2004); (b) the determination of the IMU orientation with respect to the global reference frame from gyroscopic and accelerometer data is not trivial (Woodman, 2007) and (c) in the integration of the coordinate accelerations an estimate of initial velocity needs to be provided.

Exploiting the cyclical nature of gait typically reduces the detrimental effects of the drift. This allows reduction of the interval of integration time to a single gait cycle but requires the identification in the cycle of an instant of known velocity to be used as initial velocity in the integration of the acceleration in the global reference frame. Despite the fact that during stance in level walking the foot rolls from the outer edge to the inner edge (Rodgers, 1988) and the paths of movement of the forefoot and heel differ both in shape and time (Winter, 1984), the velocity of the sensor placed on the different foot and shank locations is often set to zero at the beginning of the integration interval (zero velocity assumption— (Veltink, 2003), (Sabatini, 2005), (Foxlin, 2005), (Yun X., 2007), (Schepers, 2007), (Ojeda, 2007), (Bamberg, 2008), (Feliz, 2009), (Li, 2010), (Mariani, 2010). In particular, the foot velocity has been assumed to be zero: (a) throughout the stance phase (Yun X., 2007), (b) during a portion of it (Sabatini, 2005), (Ojeda, 2007) or (c) only in a specific instant (Li, 2010).

While the zero velocity detection issue has been recently studied (Skog, 2010)and both drift reduction and IMU orientation determination have been extensively studied (Veltink, 2003), (Foxlin, 2005), (Sabatini, 2005), (Ouinge, 2005), in literature there is no study analyzing the validity of the zero velocity assumption.

4.2. Methods

This study aimed at determining the minimum velocity of progression of various points of the foot and shank during the stance phase of the gait cycle while walking at different speeds. Such an analysis allowed estimation of the magnitude of the errors in determining the stride length due to the zero velocity assumption. The analysis of other factors affecting the stride length estimate, such as accelerometer and gyroscope drifts and biases, IMU orientation errors and identification of the time epochs when the inertial sensors are not moving, was beyond the scope of this study.

Twenty subjects (9 females, age 35.9 ± 8 years, h 168 ± 9 cm), with no history of major injuries or gross lower limb musculoskeletal abnormalities, were enrolled.

Eight retro-reflective markers (14 mm) were placed on selected locations of the right foot and shank (Fig. 1), reproducing the IMU placements adopted in various studies (Sabatini, 2005), (Schepers, 2007), (Ojeda, 2007), (Li, 2010). Subjects were asked to walk at three different self-selected speeds (slow, comfortable and fast) wearing sneaker shoes.



Figure 3 Marker locations for the right foot and shank and the global reference frame.

For three trials at each speed, marker positions were reconstructed using a stereophotogrammetric system (six-camera Vicon T20, 2Mpixel). The frame rate was set to 100 frames/s to preserve signal power (Antonsson, 1985). The measurement volume was a 1.5 m sided cube. The global reference frame was formed by a vertical axis (V), by an anteroposterior axis (AP) coincident with the direction of progression of gait and an axis perpendicular to the V and AP axes (Figure 3) The beginning and the end of the stance phase were identified, setting a threshold of 5 N to the vertical component of the ground reaction force measured with a 6-channel force platform (AMTI). For each trial, gait speed (s), gait cycle duration (T) and stride length (SL) were determined from marker positions. Trials were then reorganized in three equally populated groups (slow_s, comf_s, fast_s) based on the s values. The velocity of markers along AP (v) was estimated by applying a finite difference calculation of the relevant coordinates. At the times when the number of cameras used for reconstructing marker position changed due to loss/gain of marker visibility, abrupt artificial marker velocity changes were observed. To remove such outliers, for each marker, at any instant t_i , $v(t_i)$ was considered reliable only if both the differences $|v(t_i) - v(t_{i-1})|$ and $|v(t_{i+1}) - v(t_i)|$ were smaller than 0.005 m/s (0.015 m/s for the shank markers).

The accuracy of v estimates was assessed from the acquisition of a still marker positioned on the ground.

For each trial and each marker the minimum AP velocity during stance (v_{min}) and the relevant instant of occurrence, expressed as percentage of the stance phase $(SP_{\psi v_{min}})$, were

determined. When v reached negative numbers, v_{min} was set to zero and the relevant $SP_{\%v_{min}}$ was set to the instant of zero crossing. For each walking speed group, the average of the v_{min} values (\bar{v}_{min}) and the average of the $SP_{\%v_{min}}$ values ($\overline{SP_{\%v_{min}}}$) along with their standard deviations (sd) were computed for all markers.

From noise-free coordinate acceleration data, SL can be determined by double integrating the AP coordinate acceleration a(t) over T:

$$\int_{t=0}^{T}\int_{\tau=0}^{t}\mathbf{a}(\tau)\mathrm{d}\tau\mathrm{d}t + v_{b}T;$$

where v_b is the AP velocity at the beginning of the integration interval. Under the zero velocity assumption, the stride length estimate (SL_0) is obtained when the integration interval begins at an instant when $v_b=0$. To quantify the influence of the zero velocity assumption on the *SL* estimation, v_b was set equal to the v_{min} value found for each trial, a stride length estimate (SL^*) was obtained and the stride length estimation error (e) was calculated as the difference between SL_0 and SL^* $(e \ge SL_0 SL^* = -v_{min}T)$.

The average percent error $(\overline{e_{\%}})$ was computed for each marker and for each walking speed group as follows:

$$\overline{e_{\%}} = \frac{1}{N} \sum_{j=1}^{N} \frac{-v_{min}T_j}{SL_j} \times 100 = \frac{1}{N} \sum_{j=1}^{N} \frac{e_j}{SL_j} \times 100,$$

where *N* represents the number of trials included in each walking speed group.

4.3. Results

The error in estimating the AP velocity of a still marker was lower than 0.002 m/s.

	Stride duration s	Stride Lenght mm	Gait speed mm/s
<i>slow</i> _s	1.3 (0.2)	1122 (118)	846 (106)
<i>comf</i> _s	1.1 (0.2)	1300 (165)	1164 (99)
fast _s	0.9 (0.2)	1324 (345)	1576 (192)

Table 1 Average and sd values of stride duration (T), stride length (SL) and gait speed (s) for each walking speed group (N=20).

Gait trials were reorganized in three equally populated groups (60 trials per group) based on the effective *s* value (*slow_s*, *s* < 1 m/s; comf_s, 1 < *s* < 1.33 m/s; fast_s, *s* > 1.33 m/s). Average values of gait speed, stride length and stride duration and relevant *sd* for each walking speed group are reported in Table 1.

Minimum velocity $\overline{v_{min}}$, percentage errors $\overline{e_{\%}}$ and their *sd* are reported for each marker and for each walking speed group in Figure 4a and b, respectively. For each marker and for each walking speed group, $\overline{SP_{\%v_{min}}}$ values and *sd* are reported in Figure 5.



Figure 4 Average and sd of the minimum velocity v_{min} (a) and of the percent stride length error $e_{\%}$ (b) for each marker location and speed group.



Figure 5 Average and sd of the instant of minimum velocity v_{min} , expressed as percentage of the stance phase SP% v_{min} for each marker location and speed group.

4.4. Discussion

To determine the stride length in level walking with an IMU positioned on the foot (or the shank) a double integration of the IMU's coordinate acceleration is needed and, consequently, a velocity value at the beginning of the integration interval must be specified. In this context, velocities of foot and shank have been assumed, with no exception, to be equal to zero sometimes during the stance phase regardless of the sensor location and the gait speed. In this study, the validity of such assumption was tested by evaluating its isolated effects on stride length estimation for various IMU positions. Results showed, on average, that none of the tested points on foot and shank had zero velocity at any time during stance. However, the high *sd* values observed for some tested points indicate the high sensitivity of v_{min} to minor changes in foot contact mechanics. The minimum velocity of the foot points was always lower than that of the shank but still larger ($\overline{v_{min}} < 0.011$ m/s) than the estimate of velocity of a still marker ($v \le 0.002$ m/s).

For foot points, the zero velocity assumption applied to drift- free coordinate acceleration data introduces an average stride length underestimation error up to -0.7%. For a shank point (0.03 m above the ankle joint) the $\overline{v_{mun}}$ value resulted to be up to 0.049 m/s, causing stride length underestimation errors up to -3.3%. For all locations analyzed, $\overline{v_{mun}}$ resulted to be

dependent on gait speed. These results are consistent with the representation of the foot during stance as a deformable rolling body for which an increase of the gait speed would presumably cause an increase of the angular velocity. Since none of the observed points were in contact with the ground during stance, their AP velocity is expected to increase as the gait speed increases.

Higher gait speeds resulted also in larger stride length values and lower stance durations. As a consequence, the increase of the stride length estimation error $\overline{e_{\%}}$ due to the increase of the minimum velocity was partially counterbalanced by the decrease of the stance duration and by the increase of the stride length.

The instant of occurrence of the minimum velocity $(\overline{SP_{\%\nu_{min}}})$ ranged between 31% (CA2) and 57% (TOE) of the stance phase duration and no dependency on gait speed was observed, except for the TOE point.

In conclusion, the results of the study suggest that if an IMU is positioned on the calcaneus (CA1) or on the lateral aspect of the rear foot (CA2) the influence of the error in estimating the stride length, associated with the zero velocity assumption, is minimized since during stance these points showed (a) a minimum velocity, (b) a limited dependency on gait speed and (c) a limited timing variability. Overall the assumption that the velocity of the IMU can be set to zero at some point during stance may be acceptable if the IMU is attached to the foot, whereas it may cause critical errors if it is attached to the shank.

5. Virtual reality and inertial sensing in motor training: application to multiple sclerosis

This chapter was written on the basis of a preliminary study of VR+TM training for MS patients. The results of the study have been divided in three articles. The first of them is "Feasibility and acceptance of a virtual reality, treadmill and wearable inertial sensors system for gait training of individuals with Multiple Sclerosis" (Peruzzi, Cereatti, Mirelman, Della Croce) and has been submitted to IEEE Transactions on Neural Systems and Rehabilitation. The second article is about the effectiveness of this kind of gait intervention on MS patients and will be submitted to the Journal of NeuroEngineering and Rehabilitation. The third article will be on the outcomes of motor and cognitive tests, whose choice and analysis has been performed by the clinicians who collaborate with my group. The methods and results will be reported for a complete overview of the study.

5.1. Introduction

Many of the common motor impairments of MS could lead to gait disturbances and difficulty in walking (Swinnen, 2012). About 85% of patients with MS develop gait problems (Armutlu, 2001). Individuals with MS frequently show, compared with the healthy controls, a greater variability in lower limb kinematics during gait, reduced stride length and walking speed, prolonged double limb support time (Crenshaw, 2006), (Martin, 2006). Approximately 75% of individuals with MS experiences mobility problems (Swingler, 1992), (Lord, 1998), such as a reduced walking ability (Thoumie, 2005). Moreover, about the 55% of individuals with MS identify fatigue, and, consequently, functional walking endurance, as one of the major symptoms affecting their mobility (Fisk, 1994). Additionally gait impairments can lead also to an increased risk of falling (Cattaneo, 2002). It is therefore important to develop effective rehabilitation interventions to address gait, balance and endurance in individuals with MS. As mentioned in paragraph 2.3.1, conventional therapeutic interventions for patients with MS usually include muscle-strengthening exercises, gait and balance control techniques (O'Sullivan, 1988). Different studies have compared the effectiveness of taskoriented interventions to facilitation approaches to improve walking ability and balance in individuals with MS (Lord, 1998), (Wiles, 2001). Both methods have shown to improve functional mobility, walking speed and balance, demonstrating no significant differences in effectiveness between the two methods.

Locomotor training using a BWS and TM system is a task-oriented intervention that has shown to improve walking ability in individuals who have experienced neurological injuries such as spinal cord injury (Field-Fote, 2001), (Behrman, 2005), PD (Mehrholz, 2010), stroke (Laufer, 2001), (Nilsson, 2001), (Sullivan, 2002) and MS (Swinnen, 2012). This training modality allows for repetitive training of locomotion throughout a complete gait cycle.

Although training of locomotion combined with the use of VR has shown promising results on neurological populations (Paragraph 3.2.2) to my knowledge, only a single-case study on MS has been found in literature. Fulk et al. of a BWS system and TM for training gait combined with a VR-based system for balance purpose (Fulk, 2005), suggesting that a rehabilitation intervention combining the use of TM and VR were appropriate to improve walking, balance, and endurance outcomes on an individual with MS.

Cognitive impairments are also common (43 - 65%) in MS (Rao, 1991), often associated with depression (Arnett, 2008). Specifically the main cognitive affected areas include memory, executive functions and attention (Chiaravalloti, 2008). In a recent case-control study (Hamilton F, 2009) in MS, interaction between walking and attention has been investigated. Compared to healthy controls, MS subjects demonstrated slower speed and elevated swing time variability in gait performance under a dual task condition. Since daily life is often characterized by walking with a concurrent cognitive task, a treatment of gait in MS should incorporate motor as well as cognitive training, in order to optimally enhance mobility in dual tasking and reduce fall risk, with a consequent improvement on quality of life. It is our hypothesis that a rehabilitation intervention combining TM training and VR will produce changes in gait and endurance of MS individuals.

Moreover, the success of a rehabilitation intervention relies on the patient's engagement, motivation and satisfaction (Lewis, 2011) (Jovanov, 2005). VR-based rehabilitation tools applied to upper extremity training of post-stroke patients have shown a high level of acceptance. Cameirão et al. (Cameirão, 2010) used a questionnaire to assess the usability and acceptance of a VR-based neurorehabilitation system for controlling two virtual limbs in individualized tracking tasks. They evaluated the enjoyment in performing the task, the understanding and ease of the task and the subjective performance. Similar questionnaires have been used to assess enjoyments and level of challenge during VR training as well as self-confidence and demonstrated high levels of satisfaction of these systems (Lewis, 2011), (Schwickert, 2011). Numerous studies underlined the importance of motivation and participation of patients involved in VR-based training. Zimmerli and colleagues (Zimmerli, 2009) showed that augmented feedback applications for gait training (a VR environment and a Lokomat - Hocoma, AG, Switzerland), increased the subject's motivation and activity level. In a second work they showed that the presence of a virtual opponent in a VR environment produced higher participation and enjoyment of children with gait impairment (Koenig, 2008), (Brütsch, 2010). In Girone et al., (Girone, 2000) subjects with ankle disorders responded favorably to a training combining a VR environment with an ankle rehabilitation device. Ease of use of the device and perception of limited fatigue during the training resulted in high acceptance and satisfaction (Deutsch, 2001), (Deutsch, 2005).

The first aim of this study was to build a VR system using wearable inertial sensors and a VR environment for gait training on TM. The second aim was to assess the acceptance and the feasibility of using the VR-based gait training approach, combining motor and cognitive aspects, for patients with MS. Finally, the last aim was demonstrate the efficacy of the 6-weeks gait intervention with TM and VR in patients with MS, evaluating possible gains over ground immediately after the end of the training period.

5.2. Methods

5.2.1. Experimental set-up

The following equipment was used to administer the proposed gait training program: a conventional TM, a BWS, three IMU (MTx Xsens, Enschede, The Netherlands) and a Head Mounted Display (HMD - Z800 Emagin, Bellevue, WA, USA) – or alternatively a large screen. The TM allowed controlling the patient's walking speed. Patients wore the harness to guarantee a safe experimental setup. The HMD was used to deliver the specifically designed VR environment. Two IMUs were attached to the patient shoes and the data recorded during the walking trails were used to generate in real time, the motion of a pair of shoes in the VR environment. An additional IMU was placed on the patient's head to monitor its rotation in the horizontal plane.



5.2.2. VR environment

Figure 6 (a) A screen shot of the VR environment: a tree-lined road presenting obstacles and road bifurcations. The movement of the shoes reproduces in real time the patient's feet movement. (b) A positive visual feedback (green circle) is returned when the patient successfully passed an obstacle (a log) and (c) a negative visual feedback (red circle) is returned when the patient unsuccessfully negotiated an obstacle (a puddle). (d) A bifurcation as seen by the patient: the road sign shows two directions. The patient chooses a direction by turning her/his head towards it. The head rotation is captured by an IMU and a blinking arrow appears pointing at the selected direction just prior to the turn.

The software platform implementing the VR environment was based on Python (Python v2.4). The data extracted from the IMUs were streamed in real time into the VR environment at the sampling rate of 50 Hz. The VR environment was generated with the Vizard software (WorldViz, Santa Barbara, CA, USA).

The VR environment consisted in a tree-lined road presenting obstacles (puddles and logs). In addition, to evaluate decision-making, attention and problem solving, road bifurcations featuring street signs indicating both the pre-assigned destination and alternative destinations were included. As visual feedback, the subject could see her/his shoes while walking along the trial (Figure 6a).

While walking on the TM, when virtually approaching an obstacle, the patient was expected to negotiate the obstacle without colliding with it. This required motor abilities as well as cognitive function, specifically planning and information processing.

Environmental changes were also introduced as distracters in order to challenge divided attention. These included different modalities in the form of auditory (chirping birds, barking dogs, ambulance sirens, etc.) and visual stimuli (change in weather conditions or animals and vehicles crossing the walkway).

The VR environment had five levels of complexity. These were determined by the number of bifurcations, density of trees and road width. The trainer could set size, position and frequency of the obstacles in the walkway, according to the subject's needs. To promote motor learning, visual and auditory feedbacks were provided upon success or failure (Schultheis, 2001), (Fung, 2004), (Levin, 2010). A KR, expressed as the amount of passed obstacles, was shown on the display at the end of the training trial reflecting knowledge of results.

Gait replication

The identification of gait cycles was obtained from IMU pitch angle data: heel strikes and the toe off corresponded to IMU pitch angle minima and maxima values, respectively (Figure 7). The pitch angle data were used to reproduce the patient's shoes movement in the VR environment, which was made to move toward the patient's point of view depending on the

TM speed. The velocity of the patient's shoes in the VR environment during the swing phase v_{sw} was set under the assumption that when walking on a TM, the distance traversed during the stance phase Δs_{st} is equal to the distance traversed during swing Δs_{sw} :

$$v_{sw} = \frac{\Delta s_{sw}}{\Delta T_{sw}} \cong \frac{\Delta s_{st}}{\Delta T_{sw}} = \frac{v_{TM}\Delta T_{st}}{\overline{T} - \Delta T_{st}},$$

Where v_{TM} is the TM speed, ΔT_{sw} is the duration of the swing phase, which was assumed to be equal to the difference between the mean stride time \overline{T} (calculated over the previous 5 cycles) and the current stance phase time $\Delta T_{st}v_{TM}$.



Figure 7 Gait cycles were identified from IMU pitch angle data: heel strikes and toe off instants corresponded to the instants of pitch angle minima and maxima, respectively. The gray solid line represents the longitudinal axis of the IMU, whereas the black dotted line represents the pitch angle reference of the IMU defined by the foot flat on the ground.

Therefore, to replicate gait in the VR, a shoe was made to move forward when a maximum pitch value was detected and backward when a minimum pitch value was detected, therefore traversing a forward distance equal to $v_{sw}\Delta T_{sw}$ during swing and a distance equal to $v_{TM}st$ during the stance phase. ΔT_{st} and ΔT_{sw} are defined as:

$$\begin{cases} \Delta T_{sw} = n_{sw} \times \Delta t \\ \Delta T_{st} = n_{st} \times \Delta t \end{cases}$$

Obstacles and bifurcations

Obstacles (puddles or logs) along the walkway appeared at variable intervals of time, while their number, size and orientation were adjusted by the trainer on a trial-by-trial basis. Patients had to raise their foot, when encountering a log, and lengthen their step, when encountering a puddle. In both cases the duration of the swing time was used as an indirect measurement for discriminating successful tasks and an audio/visual feedback was generated (Figure 6b and c).

When in the proximity of a bifurcation, patients were instructed to express the chosen destination by turning their head accordingly. A blinking arrow pointing at the chosen direction, identified from the recording of the head mounted IMU, appeared just before encountering the bifurcation to allow a direction change if needed (Figure 6d).

If successful, a positive audio/visual feedback was provided immediately after the turn.

5.2.3. Study design and participants

This study used a repeated measures design (pre-training – Pre, post-training – Post, and follow-up at 4 weeks – F-Up), evaluating gait during single (ST) and dual task (DT) after a 6-week intervention using TM with VR in a single group of patients with MS.

Ten subjects affected by Relapsing Remitting type of MS according to McDonald et al.'s criteria (McDonald, 2001), were recruited from the Operative Unit of Neurology unit at the Sassari University Hospital and participated in this feasibility study (9 females, mean age: 44.3 ± 8.1 years). Patients had an Expanded Disability Status Scale - EDSS (Kurtzke, 1983) score between 3 and 6 and an Ambulation Index - AI (Rose, 2006), between 3 and 5. They had adequate cognitive ability to participate in the study (a Mini-Mental State Examination – MMSE (Folstein, 1975) - score of 24 or above), and had stable medical conditions with no relapses in the last 6 months prior to the study. Exclusion criteria included serious chronic medical illnesses (e.g., orthopedic, psychiatric or neurological) and severe visual deficits or depression. All participants provided an informed written consent prior to the beginning of the study.

5.2.4. Rehabilitation intervention

The VR environment required the participants to walk on a TM while processing multiple stimuli simultaneously and making decisions about obstacle negotiation. These decisions were made more difficult by a memory task, by distracters in the simulation to challenge the divided attention and by adjustment of the frequency and size of the virtual obstacles. The intervention lasted six weeks (with two sessions per week for a total of 12 sessions). Training progression was based on an earlier study protocol of intensive progressive individualized TM training with VR in patients with PD (Mirelman, 2011). In both studies, participants walked on the TM with a safety harness that prevented falls but did not provide BWS. The subjects were required to lift their legs high enough and far enough to pass the virtual obstacles. Sessions consisted of three trials of ten minutes of walking followed by five minutes of rest, for a total time of about 45 minutes (30 minutes of training and 15 minutes of resting). The Borg Rating Scale of Perceived Exertion Scale – Borg Scale (Borg, 1982) was administered at the beginning and end of each session to assess and monitor the level of exertion and fatigue.

Week	Goal	Session	Speed	Difficulty level	Hand support	Distracters	Number of bifurcations	
1	Familiarization	1	1.5/2.0 km/h	1			1	
	and adaptation	2	80%	1			1	
2	Increasing motor	3	00%	2		None	2	
	load		9070	2		none	2	
2 Increasing		5	100%		2			
	cognitive load	6	10070	3			3	
	Increasing both	7				_		
4	motor and cognitive load	8	110%			Easy		
5	Improve balance	9		4		Medium	4	
5	improve balance	10			1	wicdium		
6	Reach to the	11	120%	5		Hord	5	
0	maximal potential	12		5	0	паги	5	

Table 2 Training progression milestones per week and setting ranges.

The VR training allows for an individual progression outline, which would fit each patient and their specific abilities and difficulties, and would be based on their performance during previous trials and sessions. In order to keep a standardized training program for all patients, general progression should be kept within the setting ranges framework and according to specific weekly goals (Table 2). The patient's balance during training could be challenged by asking her/him to remove one or both hands from the handrail. The level of complexity of the VR environment was raised when the trainer considered the patient ready for a more intensive cognitive task. During the first session the TM speed was set at low values in order to let people to familiarize with the training. In the second session TM speed was set at 20% lower than the patients over-ground walking speed (Brooks, 2003). In the following sessions, the TM speed was adjusted based on the fatigue, the balance and the VR complexity levels. Also orientation, size, frequency of appearance, and shape of the obstacles were manipulated according to individual needs, trying to follow the guidelines for training progression, which gas been designed to achieve a success rate of 80% in clearing the obstacles to promote engagement and motor learning. Feedback was given to the participant in multiple ways including the scoring on the obstacle avoidance tasks and auditory and visual feedback about the motor and the cognitive performance.

5.2.5. Outcomes

- Evaluation of the VR-based system- The VR-based system, including hardware and VR environment, has been assessed in terms of usability (ease of use and safety of the system, accuracy of the user's task).
- Evaluation of setup and administration of the gait training program- The setup and the proposed training program were evaluated in terms of feasibility and acceptance.
 - Feasibility: a) number of patients completing the training program, b) number of unexpected events or accidents during the training, c) number of system crashes, d) number of uncompleted trials, e) TM speed progression associated to the fatigue, balance challenge and VR environment complexity levels.
- Acceptance: a questionnaire based on previous studies (Cameirão, 2010), (Chang, Chen, & Huang, 2011), (Zimmerli, 2009), (Girone, 2000), (Deutsch, 2001) was administered (Table 7). The questionnaire included aspects such as the ease and understanding of the task (statements 1-4), attitudes relating to the technology (statements 5-7), the subjective performance (statement 8-10) and enjoyment from the training (statements 11-14). Responses were recorded using a 5-point Likert scale with "strongly disagree", rated as a 1, to "strongly agree", rated as 5.

Gait analysis- Patients were asked to walk over-ground in the gait analysis laboratory under two conditions: 1) comfortable speed (ST), 2) while serially subtracting 3 from a pre - defined number (DT). A stereo-photogrammetric system (six-camera Vicon T20, frame rate 100 frames/s), two force platforms (AMTI OR6-7, frame rate 1000 frames/s) and the Vicon Plug-in Gait marker set have been used to assess spatial temporal gait parameters, low limb joint kinematics and kinetics (dynamics).

	Parameter	Gait cycle phase
	Ankle Flexion Angle Min deg	Double support
	Ankle Flexion Angle Max deg	Single support
	Ankle Flexion Angle Min deg	Swing
V:	Knee Flexion Angle Max deg	Double support
Kinematics	Knee Flexion Angle Min deg	Stance
	Knee Flexion Angle Max deg	Swing
	Hip Flexion Angle Max deg	Double support
	Hip Flexion Angle Min deg	Single support
	Ankle Moment Max Nm/Kg	Terminal Stance
D	Hip Moment Min Nm/Kg	Terminal Stance
Dynamics	Ankle Power Max W/Kg	Toe-Off
	Ankle Power Min W/Kg	Single Support
	Hip Power Max W/Kg	Swing
Sustial townseral	Gait speed mean m/s	-
Spatiai-temporal	Stride length mean m	-

Table 3 List of the analyzed parameters.

In Table 3 all the analyzed parameters have been reported. Three trials were evaluated for each condition. Since some of the patients exhibited bilateral impairments (Table 4), both data regarding the most and the less affected side has been reported.

- Endurance- The Six-minute walk test (6MWT) assessed endurance measured as the total distance walked in six minutes (Brooks, 2003). In this measure of walking endurance, subjects were instructed to cover as much distance as possible.
- *Disease severity of MS* will be measured using EDSS.
- *Cognition* Standardized cognitive tests will be used to assess executive function, visual processing and attention. These tests are: a) the Montreal Cognitive Assessment (MOCA)

to assess several cognitive domains (Nasreddine, 2005); b) the Trail Making Test³ (TMT) to evaluate visual search speed, scanning, speed of processing, mental flexibility, as well as executive functioning (Tombaugh, 2004) (Tombaugh, 2004); c) the Stroop Color-Word Test (STROOP)⁴ to measure selective attention, cognitive flexibility and processing speed, and it is used as a tool in the evaluation of executive functions (Howieson, 2004); d) frontal assessment battery (FAB), which includes simple tests of sequencing, behavioral inhibition, planning and frontal release signs, can be used as a screening test to elicit typical neurological and cognitive features (Dubois & Slachevsky, 2000); e) Serial Subtraction Dual Task (SSDT) performance. Subjects will be asked to walk while systematically subtracting three from a three-digit number. Performance in DT will be measured by the number of subtractions achieved (n°) and the number of mistakes made (err).

- Dynamic Stability a) The Timed Up-and-Go (TUG) test to assess the ability to perform sequence movements of functional mobility (Podsiadlo, 1991). Patients are timed as they stand up from a standard chair, walk a distance of 3 meters at a normal pace, turn, walk back and sit down; b) The Four Square Step Test (FSST), a measure for dynamic balance involving stepping over low objects and movement in four directions under time constraints, has been used to assess overground obstacle negotiation and fall risk (Blennerhassett, 2008). The time to complete the test is used as performance metric.
- Berg Balance Scale (BBS) To assess balance (Berg, 1995). It consists of 14 different balance tasks such as standing, reaching, bending, and transferring abilities, and has an overall score range from 0 (severely impaired) to 56 points (excellent). Although

 $^{^{3}}$ The TMT includes two versions: in the TMT A the targets are numbers and the test taker needs to connect them in sequential order, while, in the TMT B, the subject alternates between numbers and letters (1, A, 2, B, etc.). The goal of the test is for the subject is to finish the part A and part B as quickly as possible; the time taken to complete the test is used as the primary performance metric.

⁴ The STROOP contains a word page (the names of colors printed in black ink), a color page (rows of X's printed in colored ink) and a word-color page (the words from the first page are printed in the colors from the second page. The subject's task is to look at each sheet and move down the columns, reading words or naming the ink colors as quickly as possible, within a given time limit (45 seconds). Three scores, as well as an interference score, are generated using the number of items completed on each page, with higher scores reflecting better performance and less interference on reading ability.

primarily used with geriatric clients and individuals with stroke, the BBS is a valid measure of balance for individuals with MS.

 The Beck Depression Inventory (BDI) - is widely used to assess emotional wellbeing (Beck, 1974)

5.2.6. Statistical analysis

Cognitive and motor (dynamic stability and balance) outcomes, DBI, endurance, spatialtemporal parameters, kinematics and dynamics of hip, knee and ankle were analyzed and descriptive statistics were calculated. Changes between Pre and Post and changes between Pre and F-Up were analyzed using the Wilcoxon Signed Ranks test with a significance level of 0.05. The EDSS measurements at Follow-Up were compared to those at baseline using the Wilcoxon Signed Ranks with a significance level of 0.05. All the statistical analyses were performed using SPSS (version 21).

5.3. Results

None of the subjects had a relapse within the period of enrollment. A summary of demographic and baseline gait characteristics of the study population is shown in Table 4.

Patient	Age	Gender	Disease Duration [years]	EDSS Pre	Ambulation index	Distance Pre [m]	Gait speed Pre [m/s]	MMSE	Clinical
P01	40	F	17	5	3	403	1.1	29	moderate cerebellar ataxia left hemiparesis
P02	60	F	17	5.5	4	270	0.8	30	moderate cerebellar and proprioceptive right ataxia very mild right hemiparesis antero-posterior bilateral knee instability
P03	38	F	5	3.5	3	249	0.7	28	mild cerebellar ataxia left hemiparesis
P04	43	F	5	4	3	321.8	0.9	29	mild proprioceptive ataxia right hemiparesis
P05	34	F	5	6	5	212	0.6	29	cerebellar and proprioceptive ataxia double hemiparesis left > right
P06	50	F	18	4	3	391.6	1.1	30	ataxia mild to moderate proprioceptive mild right hemiparesis
P07	42	F	11	6	4	187.6	0.5	27	proprioceptive ataxia double hemiparesis left>right very anxious depressive syndrome
P08	48	М	15	4.5	3	372	1.0	22	cerebellar and proprioceptive ataxia double hemiparesis left>right

Table 4 Participants characteristics.

Half of patients were bilaterally impaired (five of them have the left as the most affected side). Only one subject used an assistive device (crutches) for gait. During the study, participants did not receive concurrent physiotherapy. All subjects tolerated the training sessions well, and no incidences of falling occurred. In the initial training sessions, all patients walked clinging to the handrails with reduced TM speed, and were impaired in the negotiation of obstacles. However, during the training they learned to walk under multi modal conditions and to divide their attention within the VR environment. During the initial session, indeed, TM speed was 0.53 m/s and patients had a mean of 18% errors in negotiating the virtual obstacles (as a percent of the total obstacles in the session). In the last session, five of eight patients walked without their hand on the handrails, the TM speed was 0.81 m/s and the mean error percent increased to 67%. The mean value of the EDSS improved significantly from 4.8 in Pre to 4.3 in F-Up (p = 0.04).

5.3.1. Evaluation of the VR-based system

The VR- based rehabilitation system was built according to the requirements of the rehabilitation protocol and tested during a 6-weeks gait intervention with TM in patients with MS. No crashes and adverse events or complications occurred during the entire training period. The evaluation of the timing performance of the VR-system was also performed and the latency between the real movement and the virtual output in the environment was lower than 100 milliseconds. The speed inside the VR environment was constant and set equal to v_{TM} , while the subject inevitability moved on the TM in medio-lateral direction, introducing a variability that was not reported in the VR environment. Moreover, similar considerations regard gait, which was reproduced, as described in the methods section, according to maxima and minima pitch values. Finally, a consideration should be made on the unrealistic view of the subject of her/his shoes. Nevertheless, none of these discrepancies have been reported by subjects, who perceived the environment as plausible. Indeed, four of them became familiar with the VR environment after the first session, while the remaining, needed an extended acclimation time (two more sessions), probably due to the lack of experience in video gaming.

5.3.2. Evaluation of setup and administration of the gait training program

Feasibility

All patients completed the training, except for two who dropped out after the first session for personal reasons.

TM speed progression for all training sessions and for all patients is reported in Table 5. The table also reports the VR environment complexity levels and the training sessions carried out without using the handrails.

TM speed km/h							Sessi	on no.					
		1	2	3	4	5	6	7	8	9	10	11	12
	1	2.0	2.8	2.8	2.8	2.8	2.6	2.6	2.6	2.6	2.4	2.4	2.4
	2	2.0	2.0	2.1	2.3	2.6	2.6	2.6	2.7	2.8	2.8*	2.8*	2.8*
.0	3	2.0	2.0	2.0	2.1	3.0	3.0	3.0	3.3*	2.8*	3.0**	3.0**	3.0**
its n	4	2.0	2.5	2.6	2.9	3.8	3.8	2.8*	3.2**	3.3**	3.4**	3.4**	3.7**
ıtien	5	1.5	1.7	1.8	2.2	2.8	3.0	3.0	2.9	2.8*	2.8*	2.8*	2.3**
Pa	6	2.0	3.1	3.1	3.1	3.4	3.4	3.4	3.6*	3.6*	3.0**	3.0**	3.1**
	7	2.0	1.5	1.7	1.8	2.5	2.6	2.7	2.8*	2.8*	2.8*	2.8*	2.8**
	8	2.0	2.0	2.3	2.5	2.9	2.9	3.2	3.5	3.0*	3.0*	3.0*	3.0*
					(COMP	LEXITY L	EVEL					
				1	2		3	4		5			

Table 5 Progression of the TM speed across the training sessions. The asterisks indicate trials in which patients took off one (*) or both (**) hands from the handrails. The gray tone of the cells indicates the complexity level. Bold numbers highlight the occasions when the trainer decided to lower the TM speed (when starting a higher level of complexity, or asking to remove hands from handrails or when fatigue increase was assessed as excessive.

Table 6 reports the mean (and standard deviation) of the Borg Scale scores at the beginning of each training session (B_0) and their increments at the end of the training session ($DB=B_r-B_0$) averaged over patients. The percentage of uncompleted trials due to fatigue in each training session is also reported.

	Session no.												
mean (std)	1	2	3	4	5	6	7	8	9	10	11	12	
B_0	8 (3)	8 (2)	8 (2)	8 (3)	8 (2)	9 (3)	8 (3)	8 (3)	9 (3)	9 (3)	8 (2)	9 (3)	
$\Delta \mathrm{B}$	7 (3)	5 (2)	4 (2)	3 (2)	4 (3)	3 (2)	4 (3)	3 (3)	2 (2)	3 (3)	3 (2)	3 (3)	
% incomplete	21	0	8	4	4	8	0	4	4	4	0	0	

Table 6 Average and standard deviation over subjects of the Borg Scale score at the beginning of training sessions (B0) and difference between Borg Scale scores at the end (Bf) and at the beginning of each training session (DB = Bf - B0). In the last row of the table the percentage of the incomplete trials due to fatigue is reported for each training session.

Acceptance

Table 7 reports the results of the satisfaction questionnaire. The number of patients who provided the same score to each of the statements is reported in the last columns of the table.

Statement/Score	1	2	3	4	5
1. I had no trouble understanding what to do in the training	-	1	1	1	5
2. It was easy for me to learn how to move my feet and the head in the VR	1	-	3	1	3
3. It was easy for me to learn how to pass the obstacles	1	-	3	1	3
4. The visual and the audio feedbacks were helpful	-	-	-	-	8
5. Wearing the HMD was comfortable	-	-	-	3	5
6. Wearing the IMUs was comfortable	-	-	-	1	7
7. Wearing the harness was comfortable	-	-	-	1	7
8. The exercise was simple	3	2	3	-	-
9. The exercise was not tiring	1	2	1	2	2
10. I made few mistakes	-	-	3	3	2
11. I have noticed some improvements in my daily life performing the training	1	-	3	1	3
12. I enjoyed the training	1	-	-	1	6
13. Participating to the training was important for me	-	-	1	2	5
14. I would like to participate to the training again	1	-	1	1	5

Table 7 The administered questionnaire with the responses given by the patients. The questionnaire includes statements regarding the understanding of the task, the acceptance of the technology, the subjective performance and the enjoinment of the training. Patients' responses were recorded using a 5-point Likert scale from "strongly disagree", rated as a 1, to "strongly agree", rated as 5.

5.3.3. Gait analysis

Since a patient could not walk autonomously during the gait assessment, eight (N=8) subjects were considered in the analysis of the lower limb kinematics and spatial-temporal parameters, while seven (N=7) for the dynamics.

Gait in single task

Gait speed during usual walking increased by 10.9% immediately after the training (Pre: 0.77 \pm 0.24 m/s, Post: 0.85 \pm 0.25 m/s, p = 0.12) and 15.2% after one month (Pre: 0.77 \pm 0.24 m/s, F-Up: 0.88 \pm 0.23 m/s, p = 0.07). Stride length also increased in Post (Pre: 0.96 \pm 0.19 m, Post: 1.04 \pm 0.20 m, p = 0.09) improving up to 12.1% in F-Up (Pre: 0.96 \pm 0.19 m, F-Up: 1.08 \pm 0.16 m, p = 0.03). Kinematics and kinetics in ST have not shown significant changes

after the training, except to the minimum value of the knee flexion angle during the stance phase of the most affected side, which increased by about two degrees (Pre: 0.9 ± 3.5 deg, Post: 2.8 ± 4.8 deg, p = 0.04). For what concerns the less affected side, the maximum value of the ankle plantar-flexion angle during the swing phase increased (Pre: 8.1 ± 8.3 deg, Post: 15.3 ± 8.3 deg, p = 0.02) and the maximum value of the knee flexion angle during the mid stance significantly decreased in Post (Pre: 15.2 ± 6.9 deg, Post: 12.7 ± 5.2 deg, p = 0.04) and in F-Up (Pre: 15.2 ± 6.9 deg, F-Up: 11.6 ± 5.8 deg, p = 0.05). About the dynamics, the maximum value of the flexor hip moment (Pre: 0.62 ± 0.29 Nm/kg, Post: 0.87 ± 0.50 Nm/kg, p= 0.13) and power (Pre: 1.12 ± 0.65 W/kg, Post: 1.26 ± 0.72 W/kg, p= 0.4) showed a positive trend after the training in the most-affected side.

Gait in dual task

Gait speed during DT significantly increased (Pre: 0.65 ± 0.19 m/s, Post: 0.77 ± 0.24 m/s, p = 0.04) by 17.7% after training, reaching the 25.1% at F-Up (Pre: 0.65 ± 0.19 m/s, Post: $0.82 \pm$ 0.21 m/s, p = 0.01). Stride length increased by 9.1% immediately after the intervention (Pre: 0.91 ± 0.15 m, Post: 1.00 ± 0.21 m, p = 0.05) improving up to 16.1% after one month (Pre: 0.91 ± 0.15 m, Post: 1.06 ± 0.10 m, p = 0.01). Also under this condition, the minimum value of the knee angle flexion during the stance phase increased in the most affected side (Pre: 0.4 \pm 4.1 deg, Post: 3.62 \pm 5.0 deg, p = 0.03). For what concerns the dynamics, the maximum values of the ankle moment (Pre: 1.20 ± 0.23 Nm/kg, Post: 1.28 ± 0.14 Nm/kg, p = 0.4, F-Up: 1.36 ± 0.16 Nm/kg, p = 0.06) and power increased in the most affected side (Pre: $1.25 \pm$ 0.69 W/kg, Post: 1.90 ± 1.02 W/kg, p = 0.03, F-Up: 2.14 ± 0.76 W/kg, p = 0.02). A not significant positive trend has been also reported for the ankle moment (Pre: 1.26 ± 0.21 , Post: 1.30 ± 0.25 , F-Up: 1.32 ± 0.23 Nm/kg) and power (Pre: 1.47 ± 0.61 W/kg, Post: 2.01 ± 1.20 W/kg, p = 0.18, F-Up: 2.09 ± 1.01 W/kg, p = 0.09) in the less affected side. Also in DT condition, the maximum value of the flexor hip moment (Pre: 0.56 ± 0.26 Nm/kg, Post: 0.82 ± 0.48 Nm/kg, p= 0.13) and power (Pre: 0.78 ± 0.43 W/kg, Post: 1.02 ± 0.59 W/kg, p= 0.18) of the most-affected side showed a positive trend immediately after the training.



Figure 8 Effects of the TM+VR intervention on overground gait speed and stride length under ST and DT condition.

5.3.4. Endurance, balance and obstacle negotiation

Endurance, as measured by the distance walked during six minutes, improved after training by a mean of 8% in distance walked (p = 0.12), amounting to an increase of 23 m (Table 8), while gait speed during the 6MWT increased from 0.83 m/s to 0.90 m/s. In F-Up, the distance walked increased by 24%, amounting to an increase of 71 m (p = 0.03) and gait speed improved to 1.03 m/s.

Dynamic stability and functional mobility improved after the training, as assessed by the significant decrease of time in performing the TUG test (Post: 17%, p = 0.05; F-Up: 29%, p = 0.02). An additional significant improvement of balance, assessed by the BBS, was also observed (Post: 7%, p=0.02; F-Up: 15%, p=0.02).

Test	Pre	Post	p value (Pre Post)	F-Up	p value (Pre-F-Up)	
			(110-1080)		(11e-r-op)	
6MWT [m]	301 ± 83	324 ± 77	0.12	372 ± 67	0.03	
TUG [s]	14 ± 5	12 ± 4	0.05	10 ± 3	0.02	
FSST [s]	23 ± 10	18 ± 6	0.01	14 ± 4	0.01	
BBS	41 ± 10	44 ± 12	0.02	47 ± 8	0.02	

Table 8 Average and standard deviation of the motor tests results for each evaluation.

Obstacles negotiation, expressed by the FSST, revealed that time for executing the coordination task significantly decreased by a mean of 22% (p = 0.01) after the intervention and by the 37% after one month (p = 0.01).

5.3.5. Neuropsychological tests

For what concerns the cognitive sphere, the results (Table 9), in general, are not significant even if several trends may be observed. Indeed score in FAB and MOCA slightly increased. The time spent in performing the STROOP slightly decreased, as well as the mean number of errors and time spent in performing the TMT-B, while the time spent in performing the TMT-A increased. Patients made 36% fewer mistakes on the cognitive task in F-Up compared with values in Pre on the SSDT.

In addition, the training appeared to have a positive influence on psychosocial aspects such as the level of depression as assessed by the BDI.

	Pre		Post		n valua	F-Up	n voluo	
Test	Mean	Std	Mean	Std	p value	Mean	Std	p value
TMT A [s]	80	30	85	33	0.14	96	71	0.53
TMT B [s]	142	111	138	63	0.74	129	49	0.89
FAB	15	2	15	2	0.34	16	2	0.09
MOCA	23	5	24	4	0.78	26	4	0.02
STROOP [s]	18	8	16	9	0.33	16	7	0.67
STROOP [err]	1	2	0	1	0.34	0	0	0.07
SSDT [n°]	14	7	15	7	0.33	17	3	0.21
SSDT [err]	2	1	2	2	0.52	1	1	0.39
BDI	20	14	19	12	0.57	13	8	0.18

Table 9 Average and standard deviation of the neuropsychological tests results for each evaluation.

5.4. DISCUSSION

The first aim of this study was to build a system and a VR environment so that they could be used to implement a rehabilitation protocol for gait on TM. The usability of both the system and the VR environment has been verified by their safe and intensive use, which lasted six weeks, with no adverse event. Secondly, although the discrepancies between real world and the VR environment, patients got familiar with it and had fun during the training, suggesting that the differences were negligible for our rehabilitation application.

Moreover, the evaluation of the experimental set up and administration of a novel VR based gait training program for patients with MS was conducted in this pilot study. The findings revealed that the patients tolerated the technology without major difficulties and demonstrated a high level of acceptance throughout the progression of the training program.

Eight of the nine patients were able to complete all training sessions without complications. The protocol allowed to effectively tune the TM speed based on: 1) the level of the VR environment complexity, 2) the amount of challenge of the patient's balance when asked to remove hands from handrails, and 3) the increase fatigue level after each training session. For all patients but one the trainer reduced the TM speed either when starting a higher level of complexity or at the same level of complexity. TM speed was generally kept constant or reduced when handrails were not being used. The average Borg Scale scores reported at the beginning of each session had limited variability across sessions (average B₀ between 8 and 9 and std between 2 and 3 - Table 6) signifying that, overall, training sessions begun at a similar level of fatigue. More importantly, the values of the average increments of the Borg Scale scores were almost constant across sessions (average varied between 2 and 5 except for the first session – Table 6) with a limited variability across subjects (ΔB std between 2 and 3 - Table 6), showing that an appropriate choice of the TM speed can also keep the fatigue increase at the end of the training session within a limited range. Moreover, the limited percentage of uncompleted trials due to fatigue in every training session (between 0 and 8% – Table 6), except for the first session, confirms that the tuning on TM speed and level of complexity was properly set. Both ΔB and the number of incomplete trials in the first session were higher than in the following sessions, suggesting that without reference to information regarding past sessions, the setting of the TM speed can result in excessive fatigue and higher number of incomplete trials.

The results obtained from the administration of the questionnaire revealed that the highest ratings were obtained for the usefulness of feedbacks and the acceptance of technology (Table 7). High ratings were also found for the ease and enjoyment from the use of the system and the training: all patients found the training easy to learn and most of them enjoyed it and would have liked to continue it. There was a high variability in the subjective responses relating to performance, highlighting the differences among patients in terms of task execution and perception of fatigue.

For what concerns the gait analysis, the results of this study showed that the six-week VRbased TM training was safe for the eight MS patients. They concluded the rehabilitation program without difficulties, enjoying every session and asking to continue the training after its end.

The main clinical findings demonstrate improved gait speed and stride length in ST condition between Pre and Post, as a result of the specific training (Figure 8). The intervention was directed at obstacle negotiation, which required the patients to take larger steps; hence the results demonstrate a specific training effect. The mechanism that allowed for this change probably lies in the kinematic analysis that demonstrated improved knee control during the mid stance in the most affected side. This in turn allowed for a more stable position that enabled a larger swing. Interestingly, most of the gains were maintained for one month, with some parameters even slightly improving from Post to F-Up. However, while taking into account the different starting values, the more impressive effects were reflected in the DT walk. The VR training is in fact a motor-cognitive DT training that implicitly improves gait. During the over ground DT evaluation, patients demonstrated improved spatial-temporal parameters (Figure 8) after training, and these changes are maintained and even improved in F-Up. These improvements were probably possible by increased moment and power generation in the ankle during push off that enabled the forward progression.

Moreover, the DT effect, measured as the difference in stride time duration between DT and ST, decreased from 0.15 seconds in Pre to 0.06 seconds in F-Up, demonstrating an improved ability of patients in dividing attention and coping with complex activities during gait. The fact that increased attention positively impacted on DT ability is confirmed also by the FSST (Table 8). In the present study, indeed, the serial subtraction executed by the patient during the evaluation of the intervention, as well as the task executed during the FSST, was not part of the training. Nevertheless after the VR-based TM training, patients walked faster under the DT condition, with longer strides compared with baseline and employed less time to execute the FSST, suggesting an ability to adapt the learned strategy to different tasks. Although, between-task transfer after VR-based TM training has been already shown in neurological

diseases, such as stroke (Jaffe, 2004) (Mirelman, 2009) and PD (Mirelman, 2011), this was not previously reported in patients with MS. Probably, the cognitive requirements characterizing the training in the VR environment allowed subjects to develop new movement strategies which are reflected in their natural motor behavior and that are maintained also after the end of the training.

This kind of intervention, as proved by the kinematics data, does not instruct to perform a specific walking pattern, but rather trains subjects to a more controlled gait also under DT conditions. Since mobility in everyday life frequently requires walking while performing simultaneous cognitive or motor tasks, this approach indirectly reduces the risk of falls of trained people and, consequently, positively impacts their quality of life.

Comparing the results of this study to those obtained by Mirelman et al. (Mirelman, 2011), the average overground speed gain during ST (0.11 m/s) and DT (0.17 m/s) are lower and less significant than those made by PD patients (ST: 0.12 m/s, DT: 0.21 m/s), but while in the current study the effective training time was 6 hours, in that case the training duration was 12 hours with a sample size of 20 instead of 8.

This pilot study has several limitations. Beyond the reduced sample size, the study design did not include a control group in order to unequivocally exclude the possibility that gains may have not been due to the training. The heterogeneity of the sample was another issue. Although the chosen criteria were very strict, a wild range of motor symptoms characterizes MS and, consequently, patients had different gait kinematics. Nonetheless, the results of the first study VR-based TM training applied to MS are quite promising and the more significant improvements during DT suggests that important gains were likely attributable to the VR and not to TM alone.

5.5. CONCLUSIONS

In this pilot study we evaluated the usability, feasibility and acceptance of a gait training setup including a TM and a VR environment created for gait training of patients with MS.

The results have shown a high level of feasibility and acceptance of the VR system and the gait training program.

This study is the first to examine the effects of TM with VR on the mobility of patients with MS. The results reported here are preliminary and the pilot study had a reduced and relatively homogeneous sample of people, therefore should be considered with caution. Nevertheless, the results indicate that intensive and progressive TM with VR training is viable for patients with MS and may positively affect complex gait conditions such as walking during dual tasking, endurance and obstacle negotiation. Larger scale, randomized controlled studies with long term follow-up are needed to confirm efficacy and retention of VR- based TM training on motor and cognitive aspects and quality of life of MS patients.

The training allowed for some flexibility (setting the TM speed and increasing level of complexity in the VR environment), which was shown to be highly important for providing tools for customizing sessions and engaging the patients whilst enhancing motivation and acceptance.

6. Inertial sensing in motor assessment: assessment of reduced arm swing in Parkinson's disease

This chapter was written on the basis of the work carried out in Tel Aviv, during my PhD period abroad, in the Laboratory for Gait and Neurodynamics of Ichilov Sourasky Medical Center. The preliminary results of the study have been published on several conference abstracts.

6.1. Introduction

PD is the second most common age-related, neurodegenerative disorder (Lewek, 2010). Tremor, rigidity, bradykinesia, and postural instability are hallmarks for the diagnosis of PD (Gelb, 1999), (Calne, 1992) with abnormal gait (e.g., short, shuffling steps, stooped posture, shorter stride length, reduced gait speed and arm swing (Morris, 1994). Although in PD reduced arm swing is the most frequently reported motor dysfunction (Nieuwboer, 1998) and it is associated with elevated fall risk (Wood, 2002), there have been only several studies aiming to quantify the arm swing of PD patients during walking. Most of them investigated shoulder kinematics in the sagittal plane using video-based (Knutsson, 1972), (Carpinella, 2007), (Zijlmans, 1996), (Behrman, 1998) or ultrasound-based systems (Roggendorf J, 2012). Since the total amount of arm swing during walking incorporates both elbow (Kuhtz-Buschbeck J. P., 2008) and trunk kinematics, it is important examining the trajectory of the end effector (e.g., wrist/hand) when quantifying arm swing. Lewek et al. defined the arm swing amplitude during walking as the distance traveled by the wrist in the anterior/posterior and medial/lateral directions with respect to the pelvis within a stride and measured it trough a motion analysis system (Lewek, 2010). They demonstrated a significant difference in the asymmetry of arm swing amplitude between early PD and healthy control subjects.

Wearable inertial sensors are relatively inexpensive and easier to implement in a clinical setting than video-based motion analysis and, therefore, ideal to assess the human motion of PD patients. Indeed, the same group (Huang, 2012), recently, attached accelerometers to each

forearm and collected angular accelerations, demonstrating that PD subjects have reduced bilateral coordination of arm swing during walking. Zampieri et al. (Zampieri, 2010) reported asymmetries in peak arm swing velocity, measured using wearable sensors, during the TUG test.

The main objective of this study was to explore how inertial sensors could be used to quantify arm swing movements during gait and to develop tools for analyzing the trials to be easily used by clinicians. The second aim of the study was to quantify arm swing amplitude during normal over-ground walking in patients with PD using wearable sensors and to compare it to the movement observed in healthy controls.

6.2. Methods

Eleven patients with PD (PDG - 7 females, 56.6 ± 8.0 years, mean disease duration 3.5 ± 2.0 years, Hoehn and Yahr Scale: II-III) and thirteen healthy controls (CTRG - 9 females, 52.1 ± 9.5) participated in this study. They were recruited from the Movement Disorders Unit at Tel Aviv Sourasky Medical Center in Israel. Two synchronized IMUs (Xsens, MTx, Enschede) were placed on the wrists of the subjects. Patients walked at their self-selected comfortable speed for one minute in a well-lit 25-meter corridor. Orientation data was extracted from sensors and processed by a custom-made algorithm in MatLab (see Appendix) to calculate parameters of arm swing, such as the range of motion and the asymmetry, of selected trials. Comparisons were made between left and right arms, affected and less affected arms and results were compared to data collected from a healthy control group.

6.3. Preliminary results

As reported in Figure 9a, arm swing range was significantly lower in the PD group (p=0.03). Figure 9b illustrates that, in the CTRG, there is a tendency for greater arm swing range of the dominant arm than the non-dominant arm (p=0.08), while PDG demonstrated (Figure 9c) a significantly higher degree of left-right asymmetry (p=0.04). Peak amplitude was moderately correlated to disease duration (r=0.61, p=0.05).



Figure 9 a) Comparison of the armswing amplitude of PDG and CTRG; b) arm swing range in CTRG; c) swing asymmetry in PDG between affected and not affected side arm.

6.4. Conclusions

In this study it was demonstrated that an automated algorithm based on data from wearable sensors is able to meaningfully quantify arm swing amplitude and asymmetry. In patients with PD this feature distinguishes between the more affected and less disease affected arms. Additional studies are needed to more fully evaluate clinical utility and the potential of this new approach.

7. Low cost video technology for tracking in virtual reality applications: the Microsoft Kinect

VR-based exercises have the potential to revolutionize therapy for people with neurological impairments and older adults at risk of falls. However, to immerse these individuals in in VR it is necessary to reproduce their movements in real-time within the virtual environment. Current tracking systems are based on optical, electromagnetic, mechanical, and inertial measurements. Usually in motor rehabilitation, the immersion in virtual environments is obtained through expensive optical systems, which are difficult to use, time and space requiring.

In the past years, low cost depth sensing cameras have also become commercially available, including the widely publicized Microsoft Kinect, which allows detecting human movement and pose without the use of markers or handheld devices (Lange, 2011).

The Microsoft Kinect is a motion sensing input device, using an RGB camera combined with an infrared-based 3D depth sensor, which comprises actively emitted structured infrared (IR) light and a single IR sensitive camera to estimate distance between the sensor and the environment. The Kinect sensor captures depth and color images simultaneously at a frame rate of up to 30 fps. The 3D depth accuracy of the Kinect camera has been evaluated, showing an accuracy of depth reconstruction in the order of 1-4 cm in the range of 0.5-5 m (Khoshelham, 2012).

Amongst its advantages, Kinect is inexpensive, easy to set up, and can be used in both home and clinical environments (Chang, 2012). Though the Kinect was originally designed for interactive entertainment, the features of the device, such as the easy interfacing with a variety of operating systems and measurement of the performance, have soon interested researchers from the rehabilitation field. Indeed, several studies have soon recognized the advantages of using an inexpensive depth camera, such as the Microsoft Kinect, for rehabilitation and assessment of body function. Actually, a system to calculate the kinematics of 20 joints of human body in real time mode has been recently proposed (Warade, 2012). A further study (Suma, 2011) tried to address the problem of skeletal tracking of a human body using the Microsoft Kinect device. The authors, using a bottom up approach, developed a skeleton toolkit that allows a programmer to add full-body control to games and VR application.

The same group implemented an interactive game-based rehabilitation tool, based on the Kinect, for training balance in adults after neurological injury (Lange, 2011).

Change et al. (Chang, 2011) developed a Kinect-based system to involve students with motor impairments in rehabilitation training. The system, based on upper limb tracking and gesture recognition, has been also evaluated in terms of effectiveness by the same authors.

Zhange and colleagues proposed a flexible motion tracking approach that can be used with a combination of Kinect devices, demonstrating its robustness and accuracy, and a significantly better performance in the presence of occlusions than current state-of-the-art implementations of single-sensor trackers (Zhang, 2012). Based on the Microsoft Kinect, non real-time applications providing accurate and robust tracking have been also implemented (Oikonomidis, 2011).

The Microsoft Kinect has been used successfully also for tracking applications of other clinical fields, including medical imaging (Noonan, 2011), robotics (Bimbo, 2012), (Loconsole, 2012) and home monitoring (Stone, 2011), (Satyavolu S., 2012)

7.1. Kinect-based gait tracking for VR clinical applications

In this study, a new, low cost motion tracking methodology, based on the use of the Microsoft Kinect is proposed and evaluated against optoelectronic technology. Several attempt have been already done in this direction. Dutta captured four cubes in a static position with the Microsoft Kinect and the multi-camera based system, to avoid the synchronization issue (Dutta, 2012). The 3D coordinates of the centers of the target cubes have been then compared through the RMS, indicating that the accuracy of that Kinect motion capture system would be at least an order of magnitude less than that of a optoelectronic system. In a recent work, Chang et al. validated the motion tracking capability

of the Kinect with respect to a multi-camera based system on three different motor tasks (Chang, 2012). They compared the trajectories and measured the relative latency between the outputs of the two systems, concluding that Microsoft Kinect is a promising VR neurological rehabilitation tool both for clinical settings and home environments. In another study, Clark et al, assessed the validity of the anatomical landmarks collected using of the Kinect against a multi-camera 3D motion analysis system (Clark, 2012) during three standing postural control assessment tests. Specifically, they assessed the anatomical landmark displacements calculated by the two systems, confirming the potential for the Kinect to be used in clinical screening programs for a wide range of patient populations.



Figure 10 A representation of a new methodology based on the use of Kinect for tracking gait in a VR environment

Since in our VR application we were interested in gait, we decided to track the feet. Considering that the accuracy of the Microsoft Kinect decreases with the distance from the sensor and that the Skeleton Tracker needs to see the head of the subject, we would need a large room, therefore losing in tracking accuracy. For these reason, it has been decided to implement a new methodology (INITION, London, UK), which did not consider the upper part of the body, but focusing on feet. The methodology is synthetically illustrated in Figure 10 and it is currently used in a multi-modal rehabilitation intervention, centered around the use of TM and VR, which aims to enhance mobility and reduce fall risk in a large sample of
PD and Mild Cognitive Impairment patients and elderly fallers [V-TIME - EU funded research project (European Union, 2012)].

7.1.1. Methods

A healthy subject (male, 35 years old) walked on a TM at 0.3-0.4 m/s, in six different ways: a) foot-dragging walk (Drag), b) asymmetric gait (Asym), c) avoiding vertical obstacles (Vert), d) with minimum clearance (Minclear), e) short steps (Short) f) normal gait (Normal). A Kinect for Windows® (Microsoft, USA)⁵ was placed 1m above the ground, and 0.95m frontally from the TM, so that it had an unobstructed view of the TM surface. A 6-camera (2MPixel) optoelectronic system (Vicon, UK) was used as gold standard.

Two green patches were placed on each shoe as markers for the Kinect: the smaller one $(5x3.5 \text{ cm}^2)$ was placed on the anterior part of the shoe and the second $(8x3.3 \text{ mm}^2)$ on the tarsus (Figure 11). Four retro-reflective markers (14 mm diameter) were placed on each shoe on the patches extremities.



Figure 11 Retro reflective markers and green patches locations on the foot. The yellow circles indicate the position of the centroids (RDP, RPP, LDP, LPP).

The Kinect captured the raw RGB and depth data at 30Hz, while the Vicon system captured at 100Hz. The two systems were synchronized. A standard color filter was used to isolate the green patches on the shoes, and then the 3D trajectories of the centroid of the patches were extracted and compared to those of the mid-point of the optical markers placed at the side of the patches (RDP, RPP, LDP, LPP – Figure 11). The comparison was carried out through estimation of the Root Mean Square Deviation (RMSD).

⁵ http://www.microsoft.com/en-us/kinectforwindows/

7.1.2. Preliminary results

The RMSD between the optical markers mid-point and the patches centroid is reported for each gait pattern and for each component (Table 10): vertical (V), medio-lateral (ML) and antero-posterior (AP).

		Gait																
RMSD mm		Drag	g		Asin	1		Vert		М	incle	ear		Shor	t	N	lorm	al
Marker	AP	V	ML	AP	V	ML	AP	V	ML	AP	V	ML	AP	V	ML	AP	V	М
RDP	11	2	2	14	4	3	14	2	2	15	2	2	14	3	2	13	3	2
RPP	11	2	1	12	3	2	14	4	6	15	3	1	13	4	2	14	5	2
LDP	11	3	2	20	3	3	13	5	2	16	2	2	10	3	2	14	4	2
LPP	12	4	3	15	6	3	13	3	4	16	3	3	10	4	2	15	5	2

Table 10 The RMSD between the optical markers mid-point and the patches centroid for each gait pattern and for each component

7.1.3. Discussion

The preliminary results of the study showed a limited RMSD, between 2 and 6 mm, along the V and ML directions, while the reconstruction of the centroid of the green patches along the AP direction was about 3 to 5 times less accurate with an RMSD value from 10 to 20 mm. This appears to be consistent with previous studies (Khoshelham, 2011), (Khoshelham, 2012), which have shown that the random error of depth measurements increases with the square of the distance from the sensor, reaching 4 cm at its maximum range. Moreover, Khoshelham and Elberink (Khoshelham, 2012) indicated also that the depth resolution decreases quadratically with increasing distance from the sensor, specifying that the point spacing in the depth direction (along the optical axis of the sensor) is as large as 7 cm at the maximum range of 5 meters. Moreover, among their final and general indications about the use of the Kinect for mapping applications, the authors also suggest to acquire data within 1–3 m distance to the sensor, because at larger distances the quality of the data would be degraded by the noise and by the low resolution of the depth measurements.

The research leading to these results has received funding from the European Union -Seventh Framework Programme (FP7-HEALTH-2011) under grant agreement n°278169 (V-TIME).

8. Augmented biofeedback: a tool for trunk motor control

Feedback has been defined as any sensory information that is available to an individual during or after the execution of a movement (Schmidt, 2008).

Augmented (extrinsic) feedback is sensory information about a movement provided in addition to intrinsic feedback, which is delivered through the sensory systems within the body. The external source may be a therapist or a device such as a biofeedback system. Traditionally the role of augmented sensory feedback in learning of motor skills has been considered satisfying, motivational, or informational in nature (Adams, 2001.). Augmented sensory feedback has been shown to facilitate muscle activation during the early stages of learning (Kim, 1997), (Mulder T., 1984), (Henry, 2007). The impact of feedback on motor learning varies as a function of the frequency, delay, and precision with which information is provided (Winstein, 1991).

Augmented feedback has been used in the past as a training tool that enables people to learn how to change physiological activity or behavior for the purposes of improving performance (Henry, 2007). There are two main goals for feedback motor training. One is to allow the central nervous system to re-establish appropriate sensory-motor loops under volitional control that may have been damaged by injury, disease, or surgery; the second goal is to assist in the development of greater cognitive awareness and control of a physiological process that has been previously considered "involuntary" or beyond the consciousness (Kenneth R., 1981).

8.1. Augmented feedback in neurological impairments

Augmented feedback training of balance, posture and motor control has shown to be effective in people affected by neurological impairments, such as peripheral neuropathy (Wu, 1997), vestibular loss (Tyler, 2003), (Kentala, 2003), (Dozza, 2007), stroke (Van Vliet, 2006), (Dursun, 1996), (Langhorne, 2009), (Stanton, 2011) traumatic brain injury (Guercio, 1997), (Wong, 1997), cerebral palsy (Talbot, 1981), (Nash, 1989), (Metherall, 1996), (Wooldridge, 1976), incomplete spinal cord injury (Kohlmeyer, 1996), (Brucker, 1996) and PD (Verschueren, 1997), (Mirelman, et al., 2011), (Marchese, 2000).

A systematic review on augmented feedback in motor recovery in subjects with neurological disorders found that extrinsic feedback was commonly provided in the form of biofeedback, kinetic feedback and kinematic feedback (Van Vliet, 2006). While biofeedback concerns physiological processes, both kinetic and kinematic feedback is related to movement variables measured during task performance: kinetic feedback variables may be related to force and torque, while kinematic feedback variables are usually derivatives of distance and time (e.g., displacement, velocity, movement time, trajectory straightness). Kinematic and kinetic feedback can be provided in relation to either the outcome of the movement or the movement pattern itself (Levin, 2010).

Moreover, the effects of biofeedback-assisted performance of balance and motor tasks have been explored using a variety of sensory feedbacks including visual, auditory and tactile modality (Dozza, 2005), (Dozza, 2007), (Verhoeff, 2009), (Vuillerme, 2007), (Wall & Kentala, 2005), (Dursun, 1996). Adamovich et al. (Adamovich, et al., 2004) developed a multi-feedback system for VR hand rehabilitation improving performance in patients with neurological impairments.

8.2. Inertial-based visual feedback tool for trunk control

Numerous tools aiming to improve trunk posture and control are based on inertial sensing: Wall et al., combining accelerometers and gyroscopes, developed a wearable vibrotactile feedback device based on trunk-tilt improved balance performance in healthy (Wall, 2001) and vestibular (Kentala, 2003) subjects. A system consisting of three inertial sensors and estimating the spinal curvature changes during trunk movements was shown to improve the subject's posture when feedback signals were provided (Huang, 2006). Using the inertial sensors of a smartphone, Franco et al., developed an integrated auditory-biofeedback tool estimating the 3-D orientation of the user's trunk during bipedal stance to improve balance (Franco, 2012).

In particular, feedback signals based on the distance from a specific target (error detection / error correction) have shown to facilitate learning processes (Magill, 2003). Several recent studies have developed and assessed feedback systems for motion guidance that use a control signal proportional to the error between the target and subject. In Sergi et al. (Sergi, 2008) and Kapur et al. (Kapur, 2010) the kinesthetic guidance systems is obtained employing magnetic motion tracking technologies and provide tactile feedback allowing patients to feel limb configuration errors continuously. Lieberman and Breazeal (Lieberman, 2007), using an optoelectronic system, developed a real-time wearable vibrotactile feedback suit to facilitate upper limb motor learning. The feedback, representing the difference, in terms of joint angles, between the target and the subject's motion, contributed to decrease the error and accelerated motor task-learning rate. Also inertial sensors have been widely used for tracking body segments in applications for motion guidance. A mobile virtual trainer, developed by Lee et al., is able to map the trainee's movements and return both instructions and real-time feedbacks based on inertial sensors data (Lee, 2011). Redd et al. developed a multisensory (visual, audible, or vibrotactile) feedback system for correcting gait information (Redd, 2011).

Chiari et al. developed a portable auditory feedback based-system that, encoding the signals provided by an accelerometer on the trunk into a stereo sound, improved balance (Chiari, 2005). Based on the latter study, we designed and implemented a real-time visual biofeedback tool.

The aim of the present work was to provide a preliminary evaluation of the usability and effectiveness of a real-time visual biofeedback (VBF) tool to assist the execution of specific motor tasks. The tool is designed to possibly become a component of a home-based rehabilitation system, relying on ICT systems aiming at providing real-time feedbacks for rehabilitation of PD patients [CuPiD FP7/2007-2013 - EU funded research project (Europen Union, 2011)].

68

8.2.1. Methods

The hardware of the developed tool is composed by an IMU (MTx Xsens, Enschede, The Netherlands) recording inertial signals and a PC providing visual feedback.

The IMU featuring a 3-axis accelerometer, a 3-axis gyroscope and a 3-axis magnetometer, was placed on the trainee's lower back (approximately on L5). Trunk inclination on the sagittal plane, computed on the device through an embedded Kalman filter, was recorded. The Graphical User Interface (GUI) providing the feedback has been implemented using a Python-based commercial software (Vizard 3.0, WorldViz, Santa Barbara, CA, USA).



Figure 12 The VBF is a sphere that is: a) green, if the subject is in the *Target Zone*, red b) and blue c) if trunk inclination exceeds respectively backward or forward.

The GUI guides the trainee in executing specific motor tasks, by providing a real-time visual feedback (Figure 12) of the difference (*err*) between a computer generated reference orientation (*Ref*) and the measured trunk orientation. If *err* is within a specific range (*Target Zone*), the sphere is green and the cursor is centred on the antero-posterior direction; if *err* increases, the sphere moves on the antero-posterior direction toward the *Limit Zones* getting red (excessive backward trunk inclination) or blue (excessive forward trunk inclination). The raising of the sphere indicates the subject when to lift from the chair.

Amplitudes of the *Target Zone* and of the *Limit Zones* are subject-specific and are based on a simple initial calibration phase (Chiari, 2005), (Nicolai, 2010).

The effectiveness and usability of the tool was evaluated in a preliminary experimental session, during which eight healthy subjects (5 males, 25-35 years) were asked to perform three types of exercises (Figure 13), lasting 20 seconds each, for five times:

- Sit-to-stand without VBF (S2S VBF);
- Trunk flexion (range 2°-20°) by sitting with VBF, as preparatory act to raise (PrepS2S + VBF);
- Sit-to-stand (in the range of flexion of the trunk 8°- 16°) with VBF (S2S + VBF)



Figure 13 Representation of the Ref during the phases of three motor exercises. The time window (TW), in which the subject is asked to perform the movement, varies from 2 to 6 seconds for task 1 and 3.

For every task, subjects were instructed, to try to follow the *Ref*, maintaining the slider cursor within the *Target Zone*. Before recording the trials, subjects performed a practice trial.

8.2.2. Preliminary results

All subjects were able to move the trunk reaching the *Target Zone* and keep it for the time required. The residence time in Target Zone for each motor task (Target Time Zone) is shown in Figure 14. Contrary to what happens in S2S - VBF, subjects in PrepS2S + VBF and in S2S +VBF are able to achieve and maintain the desired inclination of the trunk. The permanence in the target zone is greater in S2S + VBF compared to S2S - VBF and, generally, increases with the increase of TW, showing the effectiveness of the instrument VBF.

In this preliminary study, the effectiveness of a new tool based on the use of augmented visual biofeedback to assist motor tasks, such as sit-to-stand, in real time has been evaluated. The results showed the intuitiveness and ease of use of the VBF and its flexibility, resulting from the ability to adjust the time window TW and the amplitude of the target zone. The

instrument VBF therefore allows you to customize the motion exercises according to the residual capabilities of the patient and to the movement to train. In the future we intend to apply the tool VBF to pathological populations, in particular in subjects coming from a long-term bed rest in which the motor control of the trunk is usually compromised. Additionally, the control of the trunk in the medio-lateral direction will be added.

Acknowledgement

The research leading to these results has received funding from the European Union -Seventh Framework Programme (FP7/2007-2013) under grant agreement n°288516 (CuPiD project).

			Time Tar	get Zoi	ne [%]		
	Pre [s]		т	W [s]			Post[s]
	0 - 5	2	3	4	5	6	20-(5+TW)
S1	36	2	0	0	0	0	15
S2	100	0	0	0	1	0	19
S3	54	2	0	1	0	0	21
S4	100	2	2	1	1	1	41
S5	92	14	8	6	4	2	66
S6	94	8	4	3	4	1	61
S7	69	0	0	0	0	0	33
S8	100	8	4	3	3	2	68
					-		
			Time Ta	arget Z	one [9	%]	
	Pre	[s]					Post[s]
	0 -	5	5 - 10	1	LO - 15		15 - 20
S	1 81		40		24		31
S	2 95	;	31		17		19
S	3 93		40		30		21
S	4 10	0	56		35		41
S	5 10	0	74		73		66
S	6 10	0	76		68		61
S	7 86	5	32		28		33
S	8 10	0	72		65		68
			Time Tai	rget Zo	ne [%]		
	Pre [s]		Т	W [s]			Post[s]
	0 - 5	2	3	4	5	6	20-(5+TW)
S1	79	1	6	17	17	6	41
S2	88	8	8	18	13	13	34
S3	91	0	15	33	36	42	57
S4	100	26	37	47	50	30	43
S5	100	3	39	64	61	76	64
S6	100	16	64	71	67	80	45
S7	86	0	11	8	3	5	51
S8	100	39	27	59	79	65	76

Figure 14 Values of Time Target Zone for each motor task: a) S2S - VBF b) PrepS2S VBF + c) + S2S VBF. In b) has been showed the average of the Target Time Zone in the four phases of the year (Pre, I, II, Post). In a) and c) the averages are reported only for the pre-and post phases, while the phase of execution of the movement the Time Target Zone is estimated to vary from TW.

9. Conclusions

The tools and methods presented in this thesis are all based on new technologies, such as miniature inertial sensors and low-cost video systems, and have, as primary aim, the inclusion of a quantitative approach in the rehabilitation cycle that takes place in the clinical practice. A methodology based on the use of IMUs for the estimation of the arm swing in PD patients was proposed and applied in a pilot study. More needs to be done to elevate the method at the level of a clinical tool.

The use of VR and augmented feedback in rehabilitation aims at overcoming the lack of engagement and motivation of patients involved in conventional physical therapies, while controlling the specifics of the training administered. The preliminary results about the usability and flexibility of a newly developed augmented feedback tool for trunk control based on the use of IMUs are presented. The tool will be adapted to be used in training the motor control of the trunk in neurologic patients after a period of bed rest.

VR allows the rehabilitation in dual task (motor and cognitive task at the same time, typical of daily life), crucial in neurological impairments, and the transfer of the acquired skills to the real life. In this work we have shown how a VR-based intervention is effective in gait rehabilitation of MS patients, especially in dual task. Nevertheless, to ascribe these gains solely to the VR-based training, it is necessary to include a control group in the study.

Finally, two tracking systems for immersive environments have been explored in this work. We have used both non vision-based systems, such as inertial sensors, and low cost video technologies, such as the Microsoft Kinect. Inertial sensors are small, accurate, flexible, portable, but show a drift during extended measurements. This particular issue has been analyzed in this work and solutions proposed. Vision-based systems, like the Kinect®, are low-cost, flexible, easy to set up, but less accurate than non vision-based systems. Nevertheless, the preliminary results of a validation study show a limited error within three meters from the sensor, therefore acceptable for tracking in VR rehabilitation. In the near future, a more complete validation analysis will be carried out.

Appendix

I. Armswing GUI

- Once the file Armswing.fig has been opened, the GUI will be displayed on the screen.
- In the "Select a file" panel, load the right arm file and its name will be displayed in the string below while the data contained in it will appear in a table on the right side. Load also the left arm file and, eventually, the back file.

- Select a file													Exit
D: Documente)DhDIDDAT&InstignteIn111h		Counter	Acc_X	Acc_Y	Acc_Z	Gyr_X	Gyr_Y	Gyr_Z	Mag_X	Mag_Y	Mag_Z	Roll	
ase_right_S11.bd	1	32305	-0.5698	0.8338	8.7131	0.4829	-0.9865	1.3948	-0.6955	-0.0499	-0.3275	1.6196	-
	2	32306	-0.4128	0.6872	8.7205	0.4959	-0.9694	1.2635	-0.6969	-0.0413	-0.3205	1.9327	
Load left arm file	3	32307	-0.3201	0.6815	8.7498	0.4792	-0.9665	1.1164	-0.6990	-0.0359	-0.3149	2.2138	
3	4	32308	-0.2555	0.7164	8.6815	0.5064	-0.9680	0.9602	-0.7001	-0.0292	-0.3076	2.5153	
	5	32309	-0.4182	0.6479	8.3203	0.5882	-1.0015	0.7674	-0.7024	-0.0256	-0.2995	2.8463	
	6	32310	-0.5720	0.5537	7.9765	0.6274	-0.9885	0.5944	-0.7048	-0.0227	-0.2922	3.2069	
Load back file	7	32311	-0.5615	0.5662	7.6883	0.6792	-0.9683	0.4192	-0.7045	-0.0223	-0.2834	3.5864	
	8	32312	-0.4930	0.6459	7.5143	0.7456	-0.9712	0.2624	-0.7074	-0.0208	-0.2763	4.0133	
	9	32313	-0.4111	0.8050	7.3371	0.8203	-0.9478	0.1359	-0.7068	-0.0225	-0.2684	4.4755	
	10	32314	-0.2538	1.0326	7.2518	0.8941	-0.9051	0.0630	-0.7096	-0.0243	-0.2620	4.9908	
Signal	11	32315	-0.0329	1.2685	7.3867	0.9460	-0.8889	0.0370	-0.7098	-0.0270	-0.2533	5.5284	
angi tur	12	32316	0.2405	1.5329	7.6315	0.9402	-0.8399	0.0565	-0.7109	-0.0282	-0.2479	6.0713	
Orientation	13	32317	0.5182	1.7069	7.8846	0.8989	-0.8314	0.1615	-0.7113	-0.0295	-0.2412	6.5774	
	14	32318	0.7191	1.7940	8.1125	0.8227	-0.7811	0.3279	-0.7126	-0.0293	-0.2356	7.0417	
Show graphs O Acceleration	15	32319	0.7258	1.7375	8.2287	0.7325	-0.7011	0.5271	-0.7132	-0.0265	-0.2298	7.4331	
	16	32320	0.5793	1.4813	8.3419	0.6129	-0.6452	0.6681	-0.7115	-0.0228	-0.2236	7.7560	
Angular Velocity		•											•

• When all the files loaded are displayed in the strings, push the button "Show graphs" in the panel "Signal" and you will see the orientation graphs of the two arms in the right side of the GUI.



• If you want to see also acceleration and angular velocity graphs you can switch them in the panel "Signal".

- In the "Cut" panel you can cut your signal, writing the time of the start and of the end and pushing the button "Upgrade".
- If you also chose a back file you can also see the "Back's orientation" button in order to display also the back's graphs.
- Pushing the "Process and Save" button you will display the "Amplitude Table" and saving data in an excel file that you will find in the folder of the patient analyzed. The Amplitude data contains the average and the standard deviation of the ROM for every arm, the max and min values and the number of peaks.
- If the number of picks (N.picks) of the two arms (last row) is the same, the "Amplitude distribution" button will appear below. After pushed it, the amplitude distribution graph will be showed.

	Right	Left
Average	15.8308	34.7568
Std Dev	4.1223	3.6407
Maximum	24.5087	45.0630
Minimum	9.2181	27.2118
N.picks	104	104

- Once the analysis was accomplished, push the "Exit" button to close the window.
- You will find the excel file in the folder called as the patient's name of the analyzed trial.

II. Trials Analysis GUI

- Once the file TrialsAnalysis.fig has been opened, the GUI will be displayed on the screen.
- In the "Gait" panel, choose the gait modality to analyze and all the trials will appear in the list box on the right.
- You can decide to analyze all the data displayed ("Select all") or to choose only some of them. Once the file has been chosen push the "Analysis button".
- A histogram will show you the mean values and relative standard deviations of the range of amplitude of both arms for the chosen subjects. On the right a table containing the processed data will appear. Everything will be saved in an excel file.



- Once the analysis was accomplished, push the "Exit" button to close the window.
- You will find the excel file in the folder called as the chosen gait modality.

Bibliography

Adamovich, S., Merians, A., Boian, R., Tremaine, M., Burdea, G., Recce, M., et al. (2004). A virtual reality based exercise system for hand rehabilitation post-stroke: transfer to function. *26th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (IEMBS '04).*, *2*, pp. 4936-4939.

Adams, J. (2001.). Historical review and appraisal of research on the learning, retention, and transfer of human motor skills. *Psychological Bulletin*, *101*, , 41–74.

Albin, R. L., Young, A. B., & Penney, J. B. (1989). The functional anatomy of basal ganglia disorders. *Trends in Neurosciences*, *12 (10)*, 366–375.

Altun, K., & Barshan, B. (2010). Human activity recognition using inertial/magnetic sensor units. *Human Behavior Understanding* (Vol. 6219, pp. 38–51). Istanbul, Turkey, Volume Springer-Verlag: Berlin, Germany, 22 August 2010;: Springer Berlin Heidelberg.

Aminian, K., Robert, P., Buchser, E. E., Rutschmann, B., Hayoz, D., & Depairon, M. (1999). Physical activity monitoring based on accelerometry: validation and comparison with video observation. *Medical and Biological Engineering and Computing*, *37* (3), 304-308.

Antonsson, E. K., & Mann, R. W. (1985). The frequency content of gait. *Journal of biomechanics*, 18 (1), 39–47.

Armutlu, K., Karabudak, R., & Nurlu, G. (2001). Physiotherapy approaches in the treatment of ataxic multiple sclerosis: a pilot study. *Neurorehabilitation and Neural Repair*, *15(3)*, 203–211.

Arnett, P. A., Barwick, F. H., & Beeney, J. E. (2008). Depression in multiple sclerosis: review and theoretical proposal. *Journal of the International Neuropsychological Society, 14* (5), 691–724.

Auer, T., & Pinz, A. (1999). The integration of optical and magnetic tracking for multi-user augmented reality. *Computers & Graphics*, 23 (6), 805-808.

Ayrulu-Erdem, B., & Barshan, B. (2011). Leg motion classification with artificial neural networks using wavelet-based features of gyroscope signals. *Sensors*, *11*, 1721–1743.

Baillot, Y., Davis, L., & Rolland, J. (2001). A survey of tracking technology for virtual environments. In B. W. T, *Fundamentals of Wearable Computers and Augmented Reality* (pp. 67-112). Mahwah, NJ: CRC.

Baltadjieva, R., Giladi, N., Gruendlinger, L., Peretz, C., & Hausdorff, J. M. (2006). Marked alterations in the gait timing and rhythmicity of patients with de novo Parkinson's disease. *European Journal of Neuroscience, 24* (6), 1815–1820.

Bamberg, S., Benbasat, A. Y., Scarborough, D. M., Krebs, D. E., & Paradiso, J. A. (2008). Gait analysis using a shoe-integrated wireless sensor system. *IEEE Transactions on Information Technology in Biomedicine*, *12* (4), 413–423.

Bao, L., & Intille, S. (2004). Activity recognition from user-annotated acceleration data. *Pervasive Computing*, 1-17.

Beauchet, O., Dubost, V., Aminian, K., Gonthier, R., & Kressig, R. W. (2005). Dual-task-related gait changes in the elderly: does the type of cognitive task matter? *Journal of Motor Behaviour*, *37*, 259–264.

Beck, A. T., Rial, W. Y., & Rickels, K. (1974). Short form of depression inventory: cross-validation. *Psychological Reports*, *34*, 1184-1186.

Behrman, A. L., Lawless-Dixon, A. R., Davis, S. B., Bowden, M. G., Nair, P., Phadke, C., ...
& Harkema, S. J. (2005). Locomotor training progression and outcomes after incomplete spinal cord injury. *Physical Therapy*, 85 (12), 1356-1371.

Behrman, A. L., Teitelbaum, P., & Cauraugh, J. H. (1998). Verbal instructional sets to normalise the temporal and spatial gait variables in Parkinson's disease. *Journal of Neurology, Neurosurgery & Psychiatry*, 65 (4), 580–582.

Benedetti, M. G., Gasparroni, V., Stecchi, S., Zilioli, R., Straudi, S., & Piperno, R. (2009). Treadmill exercise in early mutiple sclerosis: a case series study. *European Journal of Physical Rehabilitation Medicine*, 45 (1), 53-59.

Berg, K., Wood-Dauphinee, S., & Williams, J. I. (1995). The Balance Scale: reliability assessment with elderly residents and patients with an acute stroke. *Scandinavian Journal of Rehabilitation Medicine*, *27*, 27–36.

Bergen, J. T. (2002). Aerobic exercise intervention improves aerobic capacity and movement initiation in Parkinson's disease patients. *NeuroRehabilitation. An Interdisciplinary Journal*, *17* (2), 161-168.

Bimbo, J., Rodriguez-Jimenez, S., Liu, H., Song, X., Burrus, N., Senerivatne, L. D., ... & Althoefer, K. (2012, September).. *IEEE International Conference on Multisensor Fusion and Integration for Intelligent Systems*, (pp. 65–70). Hamburg.

Bisson, E., Contant, B., Sveistrup, H., & Lajoie, Y. (2007). Functional Balance and Dual-task Reaction Times in Older Adults Are Improved by Virtual Reality and Biofeedback Training. *Cyberpsychology & Behavior : the Impact of the Internet, Multimedia and Virtual Reality on Behavior and Society, 10* (1), 16–23.

Blennerhassett, J. M., & Jayalath, V. M. (2008). The Four Square Step Test is a feasible and valid clinical test of dynamic standing balance for use in ambulant people poststroke. *Archives of physical medicine and rehabilitation, 89 (11)*, 2156–2161.

Bloem, B. R., Valkenburg, V. V., Slabbekoorn, M., & Willemsen, M. D. (2001). The Multiple Tasks Test: development and normal strategies. *Gait & Posture*, *14* (*3*), 191–202.

Bonato, P. (2009). Clinical applications of wearable technology. *Engineering in Medicine and Biology Society. EMBC 2009. Annual International Conference of the IEEE* (pp. 6580-6583). IEEE.

Bonato, P. (2010). Wearable sensors and systems. *Engineering in Medicine and Biology Magazine, IEEE, 29* (3), 25-36.

Bond, J. M. (2000). Goal-directed secondary motor tasks: their effects on gait in subjects with Parkinson disease. *Archives of physical medicine and rehabilitation*, 81(1), 110-116.

Boonstra, T. A., van der Kooij, H., Munneke, M., & Bloem, B. R. (2008). Gait disorders and balance disturbances in Parkinson's disease: clinical update and pathophysiology. *Current opinion in neurology*, *21* (4), 461-471.

Borg, G. (1982). Psychophysical bases of perceived exertion. *Medicine and Science in Sports* and *Exercise*, 14 (5), 377–381.

Bourke, A. K., O'brien, J. V., & Lyons, G. M. (2007). Evaluation of a threshold-based triaxial accelerometer fall detection algorithm. *Gait & Posture*, *26* (2), 194-199.

Bowden, M. G., Balasubramanian, C. K., Neptune, R. R., & Kautz, S. A. (2006). Anteriorposterior ground reaction forces as a measure of paretic leg contribution in hemiparetic walking. *Stroke*, *37* (3), 872-876.

Brooks, D., & Solway, S. (2003). ATS statement: guidelines for the six-minute walk test. *American Journal of Respiratory and Critical Care Medicine*, *167* (9), 1287.

Brown, T. R., & Kraft, G. H. (2005). Exercise and rehabilitation for individuals with multiple sclerosis. *Physical Medicine & Rehabilitation Clinics of North America*, *16* (2), 513-55.

Brucker, B. S., & Buylaeva, N. V. (1996). Biofeedback effect on electromyography responses in patients with spinal cord injury. *Archives of Physical Medicine and Rehabilitation*, 77:, 133-137.

Brütsch, K., Schuler, T., Koenig, A., Zimmerli, L., Mérillat, S., Lünenburger, L., ... & Meyer-Heim, A. (2010). Research Influence of virtual reality soccer game on walking

performance in robotic assisted gait training for children. *Journal of Neuroengineering and Rehabilitation*, 7 (15).

Burdea, G. (2002). Keynote Address: Virtual Rehabilitation: Benefits and and challenges. *1st Int. Workshop on Virtual Reality Rehabilitation(Mental Health, Neurological, Physical, Vocational)*,, (pp. 1–11). Lausanne, Switzerland.

Caballol, N., Martí, M. J., & Tolosa, E. (2007). Cognitive dysfunction and dementia in Parkinson disease. *Movement Disorders, 22* (17), S358–S366.

Calne, D. B., Snow, B. J., & Lee, C. (2004). Criteria for diagnosing Parkinson's disease. *Annals of neurology*, 32(S1), S125-S127.

Cameirão, M. S., Badia, S. B., Oller, E. D., & Verschure, P. F. M. J. (2010). Neurorehabilitation using the virtual reality based Rehabilitation Gaming System: methodology, design, psychometrics, usability and validation. *Journal of Neuroengineering and Rehabilitation*, 7 (1), 48.

Camicioli, R., Oken, B. S., Sexton, G., Kaye, J. A., & Nutt, J. G. (1998). Verbal fluency task affects gait in Parkinson's disease with motor freezing. *Journal of geriatric psychiatry and neurology*, *11* (4), 181.

Canning, C. G., Ada, L., & Woodhouse, E. (2008). Multiple-task walking training in people with mild to moderate Parkinson's disease: a pilot study. *Clinical Rehabilitation*, 22(3), 226-233.

Carpinella, I., Crenna, P., Marzegan, A., Rabuffetti, M., Rizzone, M., Lopiano, L., & Ferrarin, M. (2007). Effect of L-dopa and subthalamic nucleus stimulation on arm and leg swing during gait in Parkinson's disease. *IEEE Engineering in Medicine & Biology Society*, (pp. 6665–6668).

Catalfamo, P., Ghoussayni, S., & Ewins, D. (2010). Gait event detection on level ground and incline walking using a rate gyroscope. *Sensors*, *10*, 5683–5702.

Cattaneo, D., De Nuzzo, C., Fascia, T., Macalli, M., Pisoni, I., & Cardini, R. (2002). Risks of falls in subjects with multiple sclerosis. *Archives of Physical Medicine and Rehabilitation*, *83* (6), 864-867.

Chang, C.-Y., Lange, B., Zhang, M., Koenig, S., Requeio, P., Somboon, N., et al. (2012). Towards Pervasive Physical Rehabilitation Using Microsoft Kinect. *6th International Conference on Pervasive Computing Technologies for Healthcare*, (pp. 2–5). Chang, Y.-J., Chen, S., & Huang, J.-D. (2011). A kinect-based system for physical rehabilitation: A pilot study for young adults with motor disabilities. *Research in Developmental Disabilities*, *32* (6), 2566–2570.

Chaudhuri, K. R., Healy, D. G., & Schapira, A. H. (2006). Non-motor symptoms of Parkinson's disease. Diagnosis and management. *The Lancet Neurology*, *5*, 235-245.

Chen, H. C., Schultz, A. B., Ashton-Miller, J. A., Giordani, B., Alexander, N. B., & Guire, K. E. (1996). Stepping over obstacles: dividing attention impairs performance of old more than young adults. *The Journals of Gerontology Series A: Biological Sciences and Medical Sciences*, *51* (3), M116–M122.

scholChiaravalloti, N. D., & DeLuca, J. (2008). Cognitive impairment in multiple sclerosis. *Lancet Neurol*, 7 (12), 1139–1151.

Chiari, L., Dozza, M., Cappello, A., Horak, F. B., Macellari, V., & Giansanti, D. (2005). Audio-biofeedback for balance improvement: an accelerometry-based system. *IEEE Transactions on Biomedical Engineering*, *52*, 2108–2111.

Choi, J. H., Cho, J., Park, J. H., Eun, J. M., & Kim, M. S. (2008). An efficient gait phase detection device based on magnetic sensor array. *4th Kuala Lumpur International Conference on Biomedical Engineering*, *21*, pp. 778–781. Kuala Lumpur, Malaysia,.

Clark, R. A., Pua, Y. H., Fortin, K., Ritchie, C., Webster, K. E., Denehy, L., & Bryant, A. L. (2012). Validity of the Microsoft Kinect for assessment of postural control. *Gait & Posture*, *36*, 372–377.

Coates, G. (1992). Program from Invisible Site-a virtual sho, a multimedia performance work presented by George Coates Performance Works. San Francisco, CA.

Coley, B., Najafi, B., Paraschiv-Ionescu, A., & Aminian, K. (2005). Stair climbing detection during daily physical activity using a miniature gyroscope. *Gait & Posture*, *22 (4)*, 287–295.

Collin, C., Wade, D. T., Davies, S., & Horne, V. (1988). The Barthel ADL Index: a reliability study. *Disability & Rehabilitation*, *10 (2)*, 61-63.

Colombo, R., Pisano, F., Mazzone, A., Delconte, C., Micera, S., Carrozza, M. C., ... & Minuco, G. (2007). Design strategies to improve patient motivation during robot-aided rehabilitation. *Journal of NeuroEngineering and Rehabilitation*, *4* (3), 1-12.

Crenshaw, S. J., Royer, T. D., Richards, J. G., & Hudson, D. J. (2006). Gait variability in people with multiple sclerosis. *Multiple sclerosis*, *12* (5), 613-619.

D'Esposito, M., Onishi, K., Thompson, H., Robinson, K., Armstrong, C., Grossman, M. (1996). Working memory impairment in multiple sclerosis: evidence from a dual task paradigm. *Neuropsychology*, *10*, 51–56.

Dai, R., Stein, R. B., Andrews, B. J., James, K. B., & Wieler, M. (1996). Application of tilt sensors in functional electrical stimulation. *Rehabilitation Engineering, IEEE Transactions on*, *4*, 63–72.

Deutsch, J. A., Lewis, J. A., Whitworth, E., Boian, R., Burdea, G., & Tremaine, M. (2005). Formative evaluation and preliminary findings of a virtual reality telerehabilitation system for the lower extremity. *Presence: Teleoperators & Virtual Environments*, 14(2), 198–213.

Deutsch, J. E., Latonio, J., Burdea, G. C., & Boian, R. (2001). Post-Stroke Rehabilitation with the Rutgers Ankle System: a case study. *Presence: Teleoperators & Virtual Environments*, 10 (4), 416–430.

Deutscher, J. B. (2000, June). Articulated body motion capture by annealed particle filtering. *Computer Vision and Pattern Recognition*, 13-15.

Djuric, Z. (2000). Mechanisms of noise sources in microelectromechanical systems. *Microelectronic Reliability*, 40, 919–932.

Dodds, T. A., Martin, D. P., Stolov, W. C., & Deyo, R. A. (1993). A validation of the functional independence measurement and its perform- ance among rehabilitation inpatients. *Archives of Physical Medicine and Rehabilitation*, *74*, 531–536.

Doughty, K., Lewis, R., & McIntosh, A. (2000). The design of a practical and reliable fall detector for community and institutional telecare. *Journal of Telemedicine and Telecare*, *6*, 150-154.

Dozza M., Chiari, L. (2005). Audio-Biofeedback Improves Balance in Patients With Bilateral Vestibular Loss. *Archives of Physical Medicine and Rehabilitation*, *86* (7), 1401-1403.

Dozza, M. (2007). *Biofeedback systems for human postural control*. Bologna,, Italy: Ph.D. thesis, University of Bologna.

Dozza, M., Wall, C. I., Peterka, R., Chiari, L., & F. B., H. (2007). Effects of practicing tandem gait with and without vibrotactile biofeedback in subjects with unilateral vestibular loss. *Journal of Vestibular Research*, *17*:, 195-204.

Dubois, B., & Pillon, B. (1996). Cognitive deficits in Parkinson's disease. *Journal of Neurology*, 244, 2–8.

Dubois, B., & Slachevsky, A. L. (2000). The FAB: a Frontal Assessment Battery at bedside. *Neurology*, *55* (11), 1621–1626.

Dubost, V. K. (2006). Relationships between dual-task related changes in stride velocity and stride time variability in healthy older adults. *Human Movement Science*, *25* (3), 372–382.

Dursun, E., Hamamci, N., Dönmez, S., Tüzünalp, Ö., & Çakci, A. (1996). Angular biofeedback device for sitting balance of stroke patients. *Stroke*, *27*, 1354-1357.

Dutta, T. (2012). Evaluation of the Kinect sensor for 3-D kinematic measurement in the workplace. *Applied Ergonomics*, 43, 645–649.

Ebersbach, G., Milan, R. D., & Poewe, W. (1995). Influence of concurrent tasks on gait: a dual-task approach. *Perceptual and motor skills*, *81* (1), 107–113.

European Multiple Sclerosis Platform and Rehabilitation in Multiple Sclerosis. (2004). *Recommendations on Rehabilitation Services for Persons with Multiple Sclerosis in Europe*. Brussels: European Code of Good Practice in Multiple Sclerosis.

European Union. (n.d.). Retrieved Jenuary 2013, from http://www.v-time.eu/

Europen Union. (2012). Retrieved January 15, 2013, from http://www.cupid-project.eu/

Fahn, S. E. (1987). Members of the UPDRS development committee. In S. M. Fahn, *Unified Parkinson's disease rating scale. IRecent Developments in Parkinson's Disease* (pp. 153–163). Florham Park, NJ: Macmillan Health Care Information.

Faulkner, K. A., Redfern, M. S., Cauley, J. A., Landsittel, D. P., Studenski, S. A., Rosano, C.,
... & Newman, A. B. (2007). Multitasking: association between poorer performance and a history of recurrent falls. *Journal of the American Geriatrics Society*, 55(4), 570-576.

Feliz Alonso, R., Zalama Casanova, E., & Gómez García-Bermejo, J. (2009). Pedestrian tracking using inertial sensors. *Journal of Physical Agents*, *3*, 35–42.

Field-Fote, E. C. (2001). Combined use of body weight support, functional electric stimulation, and treadmill training to improve walking ability in individuals with chronic incomplete spinal cord injury. *Archives of Physical Medicine and Rehabilitation*, *82*, 818-824.

Fisk, J. D., Pontefract, A., Ritvo, P. G., Archibald, C. J., & Murray, T. J. (1994). The impact of fatigue on patients with multiple sclerosis. *Physical Medicine And Rehabilitation Clinics Of North America*, *21* (2), 9-14.

Folstein, M. F., Folstein, S. E., & McHugh, P. R. (1975). Mini-Mental State. A practical method for grading the cognitive state of patients for the clinician. *Journal of Psychiatric Research*, 189–198. Pergamon Press.

Foxlin, E. (2005). Pedestrian tracking with SHOE-mounted inertial sensors. *IEEE Computer Graphics and Applications*, 25 (1), 38–46.

Franco, C., Fleury, A., Gumery, P. Y., Diot, B., Demongeot, J., & Vuillerme, N. (2012). Freedman iBalance-ABF: A Smartphone-Based Audio-Biofeedback Balance System. *IEEE Transactions on Biomedical Engineering*, *60* (1), 211-215.

Freedman, B. S. (2010). Patent No. 2010/0118123. U.S.

Fulk, G. D. (2005). Locomotor training and virtual reality-based balance training for an individual with multiple sclerosis: a case report. *Journal of neurologic physical therapy*, *29* (1), 34-42.

Fung, J., Malouin, F., McFadyen, B. J., Comeau, F., Lamontagne, A., Chapdelaine, S., ... & Richards, C. L. (2004). Locomotor rehabilitation in a complex virtual environment. *International Conference of IEEE Engineering in Medicine and Biology Society*, 7, pp. 4859– 4861.

Fung, J., Richards, C. L., Malouin, F., McFadyen, B. J., & Lamontagne, A. (2006). A Treadmill and Motion Coupled Virtual Reality System for Gait Training Post-Stroke. *CyberPsychology & Behavior*, 9 (2), 157-163.

Gao, H. M., & Hong, J. S. (2008). Why neurodegenerative diseases are progressive: uncontrolled inflammation drives disease progression. *Trends in immunology*, *29* (8), 357-365.

Gelb, D. J., Oliver, E., & Gilman, S. (1999). Diagnostic criteria for Parkinson's disease. *Archives of Neurology*, 56 (1), 33–39.

Girone, M., Burdea, G., Bouzit, M., Popescu, V., & Deutsch, J. E. (2000). Orthopedic rehabilitation using the 'Rutgers ankle' interface. *Studies In Health Technology And Informatics*, 70, 89–95.

Godfrey, A., Conway, R., Meagher, D., & ÓLaighin, G. (2008). Direct measurement of human movement by accelerometry. *Medical Engineering & Physics, 30 (10)*, 1364-1386.

Graham, D. L., Ferreira, H. A., & Freitas, P. P. (2004). Magnetoresistive-based biosensors and biochips. *Trends in Biotechnology*, *22*, 455–462.

Guercio, J., Chittum, R., & McMorrow, M. (1997Self-management in the treatment of ataxia: a case study in reducing ataxic tremor through relaxation and biofeedback. *Brain Injuries, 11* (5), 353-362.

Guo, S., Song, Z., & Ren, C. (2009, August). Development of upper limb motor function training and rehabilitation system. *IEEE International Conference on Mechatronics and Automation*, (pp. 931–936). Changchum, China.

Gutierrez, G. M., Chow, J. W., Tillman, M. D., McCoy, S. C., Castellano, V., & White, L. J. (2005). Resistance training improves gait kinematics in persons with multiple sclerosis. *Archives of physical medicine and rehabilitation*, *86* (9), 1824-1829.

Hamilton, F., Rochester, L., Paul, L., Rafferty, D., O'Leary, C. P., & Evans, J. J. (2009). Walking and talking: an investigation of cognitive-motor dual tasking in multiple sclerosis. *Multiple Sclerosis*, *15* (10), 1215–1227.

Hausdorff, J. M., Edelberg, H. K., Mitchell, S. L., Goldberger, A. L., & Wei, J. Y. (1997). Increased gait unsteadiness in community-dwelling elderly fallers. *Archives of physical medicine and rehabilitation*, 78 (3), 278-283.

Hausdorff, J. M., Balash, J., & Giladi, N. (2003). Effects of cognitive challenge on gait variability in patients with Parkinson's disease. *Journal of Geriatric Psychiatry and Neurology*, *16* (1), 53-58.

Henderson, A., Korner-Bitensky, N., & Levin, M. (2007). Virtual reality in stroke rehabilitation: a systematic review of its effectiveness for upper limb motor recovery. *Topics in stroke rehabilitation*, *14* (2), 52-61.

Henry, S. M., & Teyhen, D. S. (2007). Ultrasound imaging as a feedback tool in the rehabilitation of trunk muscle dysfunction for people with low back pain. *Journal of Orthopaedic & Sports Physical Therapy*, *37*, 627-634.

Hilgard, E. R. (1975). Hypnosis. Annual Review of Psychology, 26(1), 19-44.

Hirtz, D., Thurman, D. J., Gwinn-Hardy, K., Mohamed, M., Chaudhuri, A. R., & Zalutsky, R. (2007). How common are the "common" neurologic disorders? *Neurology*, *68* (5), 326-337.

Holden, M. K. (2005). Virtual environments for motor rehabilitation: review. *Cyberpsychology & behavior : the impact of the Internet, multimedia and virtual reality on behavior and society*, 8 (3), 187-211.

Holtzer, R., Verghese, J., Xue, X., Lipton, R.B. (2006). Cognitive processes related to gait velocity: results from the Einstein Aging Study. *Neuropsychology*, *20* (2), 215.

Horak, F. (1990). Assumptions underlying motor control for neurologic rehabilitation. In *Contemporary Management of Motor Control Problems* (pp. 11–27). Alexandria: VA: American Physical Therapy Association.

Howieson, M. L. (2004). *Neuropsychological assessment* (4th ed.). New York:: Oxford University Press.

Huang, H., Wolf, S. L., & He, J. (2006). Recent developments in biofeedback for neuromotor rehabilitation. *Journal of NeuroEngineering and Rehabilitation*, *3* (11), 1-12.

Huang, X., Mahoney, J. M., Lewis, M. M., Du, G., Piazza, S. J., & Cusumano, J. P. (2011). Both coordination and symmetry of arm swing are reduced in Parkinson's disease. *Gait & Posture*, *35*, 373–377.

Hussain, A., Roach, N., Balasubramanian, S., & Burdet, E. (2012). A Modular Sensor-Based System for the Rehabilitation and Assessment of Manipulation. *IEEE Haptics Symposium (HAPTICS)*, (pp. 247-254).

Jacobs, L. (1986). Blood lactate. Implications for training and sports performance. *Sports Medicine*, *3*, 10–25.

Jaffe, D. L., Brown, D. A., Pierson-Carey, C. D., Buckley, E. L., & Lew, H. L. (2004). Stepping over obstacles to improve walking in individuals with poststroke hemiplegia. *Journal of Rehabilitation Research and Development*, *41* (*3*), 283–292.

Jankovic, J. (2008). Parkinson's disease: clinical features and diagnosis. *Journal of Neurology, Neurosurgery & Psychiatry*, 79 (4), 368-376.

Jovanov, E., Milenkovic, A., Otto, C., & De Groen, P. C. (2005). A wireless body area network of intelligent motion sensors for computer assisted physical rehabilitation. *Journal of NeuroEngineering and Rehabilitation*, 2 (1), 6.

Kapur, P., Jensen, M., Buxbaum, L. J., Jax, S. A., & Kuchenbecker, K. J. (2010). Spatially distributed tactile feedback for kinesthetic motion guidance. *IEEE Haptics Symposium*, (pp. 519–526).

Katzman, R. B. (1983). Validation of a Short Orientation-Memory-Concentration Test of Cognitive Impairment. *American Journal of Psychiatry*, *140* (6), 734-739.

Kennedy, R. S., & Stanney, K. M. (1996). Postural instability induced by virtual reality exposure: Development of a certification protocol. *International Journal of Human* - *Computer Interaction*, 8 (1), 25-47.

Kennedy, R. S., Stamrey, K. M., Ordy, J. M., & Dunlap, W. P. (1997). Virtual reality effects produced by head-mounted display (HMD) on human eye-hand coordination, postural equilibrium, and symptoms of cybersickness. *Society for Neuroscience, 23*, 772.

Kenneth, R. Gaarder, P., Montgomery, S. (1981). *Clinical biofeedback: a procedural manual for behavioral medicine*. Baltimore, MD: Williams & Wilkins.

Kentala, E., Vivas, J., & Wall, C. (2003). Reduction of postural sway by use of a vibrotactile balance prosthesis prototype in subjects with vestibular deficits. *The Annals of otology, rhinology, and laryngology, 112* (5), 404-409.

Khoshelham, K. (2011). Accuracy analysis of kinect depth data. *ISPRS Workshop Laser Scanning*, 38.

Khoshelham, K, Elberink, S. O. (2012). Accuracy and resolution of kinect depth data for indoor mapping applications. *Sensors*, *12* (2), 1437–1454.

Kim, H. J., & Kramer, J. F. (1997). Effectiveness of visual feed- back during isokinetic exercise. *Journal of Orthopaedic & Sports Physical Therapy*, *26 (6)*, 318-323.

Knutsson, E. (1972). An analysis of parkinsonian gait. Brain, 95 (3), 475-486.

Koenig, A., Brutsch, K., Zimmerli, L., Guidali, M., Duschau-Wicke, A., Wellner, M., ... & Riener, R. (2008). Virtual environments increase participation of children with cerebral palsy in robot-aided treadmill training. *Virtual Rehabilitation*, 121–126.

Kohlmeyer, K. M., Hill, J. P., Yarkony, G. M., & Jaeger, R. J. (1996Electrical stimulation and biofeedback effect on recovery of tenodesis grasp: a controlled study. *Archives of Physical Medicine and Rehabilitation*, 77 (7), 702-706.

Kuhtz-Buschbeck, J. P., Brockmann, K., Gilster, R., Koch, A., & Stolze, H. (2008). Asymmetry of arm-swing not related to handedness. *Gait & Posture*, *27* (3), 447–454.

Kurtzke, J. F. (1983). Rating neurologic impairment in multiple sclerosis: an expanded disability status scale (EDSS). *Neurology*, *33* (11), 1444-1452.

Lange, B., Chang, C.-Y., Suma, E., Newman, B., Rizzo, A., & Bolas, M. (2011). Development and evaluation of low cost game-based balance rehabilitation tool using the microsoft kinect sensor. *International Conference of IEEE Engineering in Medicine and Biology Society*, (pp. 1831–1834).

Lange, B., Rizzo, S., Chang, C. Y., Suma, E. A., & Bolas, M. (2011, January). Markerless Full Body Tracking: Depth-Sensing Technology within virtual environments. *Interservice / Industry Training, Simulation and Education Conference - I/ITSEC*), 2011(1).

Langhorne, P., Coupar, F., & Pollock, A. (2009). Motor recovery after stroke: a systematic review. *The Lancet Neurology*, *8*, 741–754.

Laufer, Y., Dickstein, R., Chefez, Y., & Marcovitz, E. (2001). The effect of treadmill training on the ambulation of stroke survivors in the early stages of rehabilitation: a randomized study. *Journal Of Rehabilitation Research And Development, 38*, 69-78.

Laver, K.E., George, S., Thomas, S., Deutsch, J.E., Crotty, M. (2011). Virtual reality for stroke rehabilitation. *Cochrane Database of Systematic Reviews*, 9: CD008349.

Lee, R. Y., Laprade, J., & Fung, E. H. (2003). A real time gyroscopic system for three dimensional measurement of lumbar spine motion. *Medical Engineering & Physics, 25(10),* 817–824.

Lee, B. - C., Chen, S., & Sienko, K. (2011). A Wearable Device for Real-Time Motion Error Detection and Vibrotactile Instructional Cuing. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, *19* (4), 374-381.

Levin, M. F., Knaut, L. A., Magdalon, E. C., & Subramanian, S. (2009). Virtual reality environments to enhance upper limb functional recovery in patients with hemiparesis. *Studies In Health Technology And Informatics*, 145, 94–108.

Levin, M. F. (2010). Feedback and virtual environments for motor learning and rehabilitation. Schedae.

Lewek, M. D., Poole, R., Johnson, J., Halawa, O., & Huang, X. (2010). Arm swing magnitude and asymmetry during gait in the early stages of Parkinson's disease. *Gait Posture*, *2*, 256–260.

Lewis, C. A. (1998). Human factors consideration in clinical applications of virtual reality. In G. Riva, *Virtual Reality in Neuro-Psycho-Physiology*. Amsterdam: IOS Press.

Lewis, G. W. (2011). Virtual Reality Games for Rehabilitation : Perspectives from the Users and New Directions. *International Conference on Virtual Rehabilitation*, (pp. 1–2).

Lewis, G. W. (2011). Virtual reality games for rehabilitation of people with stroke: perspectives from the users. *Disability and Rehabilitation: Assistive Technology*, *6* (5), 453–463.

Li, Q., Young, M., Naing, V., & Donelan, J. M. (2010). Walking speed estimation using a shank-mounted inertial measurement unit. *Journal of Biomechanics*, *43* (8), 1640–1643.

Lidcombe. (2003). Update on medical management of multiple sclerosis to staff of the Multiple Sclerosis Society of New South Wales. *Multiple Sclerosis Society of New South Wales*.

Lieberman, J., & Breazeal, C. (2007). Development of a Wearable Vibrotactile Feedback Suit for Improved Human Motor Learning. *IEEE TRANSACTIONS ON ROBOTICS*, *23* (5), 919-926.

Lindenberger, U., Marsiske, M., & Baltes, P. B. (2000). Memorizing while walking: increase in dual-task costs from young adulthood to old age. *Psychology and aging*, *15* (3), 417.

Loconsole, C., Banno, F., Frisoli, A., & Bergamasco, M. (2012). A New Kinect-based Guidance Mode for Upper Limb Robot-aided Neurorehabilitation. *International Conference on Intelligent Robots and Systems*, (pp. 1037–1042.).

Lord, S. E., Wade, D. T., & Halligan, P. W. (1998). A comparison of two physiotherapy treatment approaches to improve walking in multiple sclerosis: a pilot randomized controlled study. *Clinical Rehabilitation*, *12* (6), 477-486.

Lorincz, K., Chen, B. R., Challen, G. W., Chowdhury, A. R., Patel, S., Bonato, P., & Welsh, M. (2009, November). Mercury: a wearable sensor network platform for high-fidelity motion analysis. In *Proceedings of the 7th ACM Conference on Embedded Networked Sensor Systems* (pp. 183-196). ACM.Lorincz.

Luinge, H. J., & Veltink, P. H. (2005). Measuring orientation of human body segments using miniature gyroscopes and accelerometers. *Medical and Biological Engineering and Computing*, 43, 273–282.

Lundin-Olsson, L. N. (1997). "Stops walking when talking" as a predictor of falls in elderly people. *The Lancet*, 349:617.

Lykouras, L., Oulis, P., Adrachta, D., Daskalopoulou, E., Kalfakis, N., Triantaphyllou, N., ... & N Christodoulou, G. N. N. (1998). Beck Depression Inventory in the Detection of Depression among Neurological Inpatients. *Psychopathology*, *31*, 213–219.

Magill, R. (2003). *Motor learning and control: Concepts and applications* (7th ed.). New York: McGraw-Hill.

Malouin, F., Richards, C. L., McFadyen, B., & Doyon, J. (2003). New perspectives of locomotor rehabilitation after stroke. *Médecine sciences*, *19*, 994-998.

Mancini, M., & Horak, F. B. (2010). The Relevance of Clinical Balance Assessment Tools to Differentiate Balance Deficits. *European Journal of Physical and Rehabilitation Medicine*, 46 (2), 239–248.

Mancini, M., Salarian, A., Carlson-Kuhta, P., Zampieri, C., King, L., Chiari, L., ... & Rymer, W. Z. (2012). ISway: a sensitive, valid and reliable measure of postural control. *Journal of NeuroEngineering and Rehabilitation*, 9 (1), 59.

Mantovani, F., Castelnuovo G. (2003). Sense of presence in virtual training: enhancing skills acquisition and transfer of knowledge through learning experience in virtual environments. In G. Riva, F. Davide, W.A IJsselsteijn, *Being There: Concepts, effects and measurement of user presence in synthetic environments.* Ios Press.

Marchese, R., Diverio, M., Zucchi, F., Lentino, C., & Abbruzzese, G. (2001). The role of sensory cues in the rehabilitation of Parkinsonian patients: a comparison of two physical therapy protocols. *Movement Disorders*, *15* (*5*), 879–883.

Mariani, B., Hoskovec, C., Rochat, S., Büla, C., Penders, J., & Aminian, K. (2010). 3D gait assessment in young and elderly subjects using foot-worn inertial sensors. *Journal of Biomechanics*, 43 (15), 2999–3006.

Martin, C. L., Phillips, B. A., Kilpatrick, T. J., Butzkueven, H., Tubridy, N., McDonald, E., & Galea, M. P. (2006). Gait and balance impairment in early multiple sclerosis in the absence of clinical disability . *Multiple sclerosis*, *12* (5), 620-628.

Mathie, M. J., Coster, A. C., Lovell, N. H., & Celler, B. G. (2004). Accelerometry: providing an integrated, practical method for long-term, ambulatory monitoring of human movement. *Physiological measurement*, *25* (2), R1.

Mayagoitia, R. E., Nene, A. V., & Veltink, P. H. (2002). Accelerometer and rate gyroscopes measurement of kinematics: an inexpensive alternative to optical motion analysis systems. *Journal of Biomechanics*, *35*, 537–542.

Mayeux, R., Stern, Y., Cote, L., & Williams, J. B. (1984). Altered serotonin metabolism in depressed patients with Parkinson's disease. *Neurology*, *34* (5), 642-642.

McDonald, W. I., Compston, A., Edan, G., Goodkin, D., Hartung, H. P., Lublin, F. D., ... & Wolinsky, J. S. (2001). Recommended diagnostic criteria for multiple sclerosis: guidelines from the International Panel on the diagnosis of multiple sclerosis. *Annals of Neurology*, *50* (1), 121–127.

Mehrholz, J., Friis, R., Kugler, J., Twork, S., Storch, A., & Pohl, M. (2010). Treadmill Training for Patients with Parkinson' s Disease. *Cochrane Database of Systematic Reviews, 1*, 1–33.

Menz, H. B., Lord, S. R., & Fitzpatrick, R. C. (2003). Acceleration patterns of the head and pelvis when walking on level and irregular surfaces. *Gait & Posture*, *18*, 35–46.

Merritt, J. (1981). Management of spasticity in spinal cord injury. *Mayo Clinical Proceedings*, 56 (10), 614–622.

Metherall, P., Dymond, E. A., & Gravill, N. (1996). Posture control using electrical stimulation biofeedback: a pilot study. *Journal of Medical Engineering & Technology, 20* (2), 53–59.

Mirelman, A., Bonato, P., & Deutsch, J. E. (2009). Effects of training with a robot-virtual reality system compared with a robot alone on the gait of individuals after stroke. *Stroke: A Journal of Cerebral Circulation*, 40 (1), 169–174.

Mirelman, A. M. (2011). Virtual reality for gait training: can it induce motor learning to enhance complex walking and reduce fall risk in patients with Parkinson's disease? *The Journals Of Gerontology. Series A*, 66 (2), 234–240.

Mirelman, A., Herman, T., Nicolai, S., Zijlstra, A., Zijlstra, W., Becker, C., et al. (2011). Audio-Biofeedback training for posture and balance in Patients with Parkinson's disease. *Journal of NeuroEngineering and Rehabilitation*, 8 (35), 1-7.

Miyai, I., Fujimoto, Y., Yamamoto, H., Ueda, Y., Saito, T., Nozaki, S., & Kang, J. (2002). Long-term effect of body weight-supported treadmill training in Parkinson's disease: a randomized controlled trial. *Archives Of Physical Medicine And Rehabilitation*, *83* (10), 1370–1373.

Miyazaki, S. (1997). Long-term unrestrained measurement of stride length and walking velocity utilizing a piezoelectric gyroscope. *IEEE Transactions on Biomedical Engineering*, *44*, 753–759.

Morris, M. E., Iansek, R., Matyas, T. A., & Summers, J. J. (1994). The pathogenesis of gait hypokinesia in Parkinson's disease. *Brain*, *117* (5), 1169–1181.

Mulder, T., & Hulstyn, W. (1984). Sensory feedback therapy and theoretical knowledge of motor control and learning. *American Journal of Physical Medicine*, 63, 226–244.

Najafi, B., Aminian, K., Loew, F., Blanc, Y., & Robert, P. A. (2002). Measurement of standsit and sit- stand transitions using a miniature gyroscope and its application in fall risk evaluation in the elderly. *IEEE Transactions on Biomedical Engineering*, 49 (8), 843–851.

Najafi, B., Aminian, K., Paraschiv-Ionescu, A., Loew, F., Bula, C. J., & Robert, P. (2003). Ambulatory system for human motion analysis using a kinematic sensor: monitoring of daily physical activity in the elderly. *IEEE Transactions on Biomedical Engineering*, *50* (6), 711–723.

Nash, J., Neilson, P. D., & O'Dwyer, N. J. (1989). Reducing spasticity to control muscle contracture of children with cerebral palsy. *Developmental Medicine & Child Neurology*, *31*, 471–480.

Nasreddine, Z. S., Phillips, N. A., Bédirian, V., Charbonneau, S., Whitehead, V., Collin, I., ... & Chertkow, H. (2005). The Montreal Cognitive Assessment, MoCA: a brief screening tool for mild cognitive impairment. *Journal of the American Geriatrics Society*, *53*, 695–699.

Nieuwboer, A., Weerdt, W. D., Dom, R., & Lesaffre, E. (1998). A frequency and correlation analysis of motor deficits in Parkinson patients. *Disability & Rehabilitation, 4*, 142–150.

Nilsson, L., Carlsson, J., Danielsson, A., Fugl-Meyer, A., Hellström, K., Kristensen, L., ... & Grimby, G. (2001). Walking training of patients with hemiparesis at an early stage after stroke: a comparison of walking training on a treadmill with body weight support and walking training on the ground. *Clinical Rehabilitation*, *15*, 515–527.

Noonan, P. J., Cootes, T. F., Hallett, W. A., & Hinz, R. (2011). The design and initial calibration of an optical tracking system using the Microsoft Kinect. *IEEE Nuclear Science Symposium and Medical Imaging Conference (NSS/MIC)* (pp. 3614-3617). IEEE.

O'Shea, S., Morris, M. E., & Iansek, R. (2002). Dual task interference during gait in people with Parkinson disease: effects of motor versus cognitive secondary tasks. *Physical Therapy*, *82* (9), 888–897.

O'Sullivan S., Schmitz T. J., editors. (1988). *Physical Rehabilitation: Assessment and Treatment*. 3rd ed. Philadelphia: FA Davis Company; 1994, p. 327–60. 3.

O'Donovan, K. J., Kamnik, R., O'Keeffe, D. T., & Lyons, G. M. (2007). An inertial and magnetic sensor based technique for joint angle measurement. *Journal of Biomechanics*.

Oberg, P. A., Togawa, T., & Spelman, F. A. (Eds.). (2006). *Sensors in Medicine and Health Care*. Berlin, Germany: Weinheim, Wiley-VCH.

Oikonomidis, I., Kyriazis, N., & Argyros, A. (2011). Efficient model-based 3D tracking of hand articulations using Kinect. 22nd British Machine Vision Conference (BMV C 2011). Dundee.

Ojeda, L., & Borenstein, J. (2007). Non-GPS navigation for security personnel and first responders. *Journal of Navigation*, *60* (3), 391–407.

Palmerini, L., Mellone, S., Avanzolini, G., Valzania, F., & Chiari, L. (2013). Quantification of Motor Impairment in Parkinson's Disease Using an Instrumented Timed Up and Go Test. *IEEE Transactions Neural Systems and Rehabilitation Engineering*.

Parkka, J., Ermes, M., Korpipaa, P., Mantyjarvi, J., Peltola, J., & Korhonen, I. (2006). Activity classification using realistic data from wearable sensors. *IEEE Transactions on Information Technology in Biomedicine*, 10 (1), 119-128.

Parnandi, A., Wade, E., & Mataric, M. (2010). Motor function assessment using wearable inertial sensors. In Engineering in Medicine and Biology Society (EMBC), 2010 *Annual International Conference of the IEEE Engineering in Medicine and Biology Society* (pp. 86–89).

Pearson, O. R., Busse, M. E., Van Deursen, R. W. M., & Wiles, C. M. (2004). Quantification of walking mobility in neurological disorders. *Quarterly Journal of Medicine*, 97 (8), 463-475.

Petajan, J. H., Gappmaier, E., White, A. T., Spencer, M. K., Mino, L., & Hicks, R. W. (1996). Impact of aerobic training on fitness and quality of life in multiple sclerosis. *Annals of neurology*, *39* (4), 432-441.

Pilutti, L. A., Lelli, D. A., Paulseth, J. E., Crome, M., Jiang, S., Rathbone, M. P., & Hicks, A. L. (2011). Effects of 12 weeks of supported treadmill training on functional ability and quality of life in progressive multiple sclerosis: a pilot study. *Archives of Physical Medicine and Rehabilitation*, *92* (1), 31-36.

Plotnik, M., Dagan, Y., Gurevich, T., Giladi, N., & Hausdorff, J. M. (2011). Effects of cognitive function on gait and dual tasking abilities in patients with Parkinson's disease suffering from motor response fluctuations. *Experimental brain research*, 208 (2), 169-179.

Plotnik, M., Giladi, N., & Hausdorff, J. M. (2008). Bilateral coordination of walking and freezing of gait in Parkinson's disease. *European Journal of Neuroscience*, 27 (8), 1999-2006.

Plummer-D'Amato, P., Kyvelidou, A., Sternad, D., Najafi, B., Villalobos, R., & Zurakowski, D. (2012). Training dual-task walking in community-dwelling adults within 1 year of stroke: a protocol for a single-blind randomized controlled trial. *BMC neurology*, *12* (1), 129.

Podsiadlo, D., & Richardson, S. (1991). The timed" Up & Go": a test of basic functional mobility for frail elderly persons. *Journal of the American Geriatrics Society, 39* (2), 142-148.

Preece, S. J., Goulermas, J. Y., Kenney, L. P., Howard, D., Meijer, K., & Crompton, R. (2009). Activity identification using body-mounted sensors—a review of classification techniques. *Physiological measurement*, *30* (4), R1.

Rand, D., Kizony, R., Feintuch, U., Katz, N., Josman, N., Rizzo, A. S., & Weiss, P. L. (2005). Comparison of two VR platforms for rehabilitation: video capture versus HMD. *Presence Teleoperators & Virtual Environments*, *14(2)*, *147-160*.

Rao, S. M., Leo, G. J., Bernardin, L., & Unverzagt, F. (1991). Cognitive dysfunction in multiple sclerosis. I. Frequency, patterns, and prediction. *Neurology*, *41* (5), 685–691.

Ravi, N., Dandekar, N., Mysore, P., & Littman, M. L. (2005, July). Activity recognition from accelerometer data. *National conference on artificial intelligence* (Vol. 20, No. 3, p. 1541). Menlo Park, CA; Cambridge, MA; London; AAAI Press; MIT Press;.

Redd, C. B., & Bamberg, S. J. M. (2011, August). A wireless sensory feedback system for real-time gait modification. In Engineering in Medicine and Biology Society, EMBC, 2011 *Annual International Conference of the IEEE* (pp. 1507-1510). IEEE.

Ribo, M., Pinz, A., & Fuhrmann, A. L. (2001). A new optical tracking system for virtual and augmented reality applications. *18th IEEE Instrumentation and Measurement Technology Conference*, *3*, pp. 1932–1936.

Riva, G., Rizzo, A., Alpini, D., Attree, E. A., Barbieri, E., Bertella, L., ... & WEISS, P. L. (1999). Virtual environments in the diagnosis, prevention, and intervention of age-related diseases: a review of VR scenarios proposed in the EC VETERAN Project. *CyberPsychology & Behavior, 2*, 577–591.

Riva, G., Davide, F., & IJsselsteijn, W. A. (2003). Being There: Concepts, effects and measurements of user presence in synthetic environments (Vol. 5). Ios Press Inc.

Rizzo, A., Buckwalter, J. G., van der Zaag, C., Neumann, U., Thiebaux, M., Chua, C., ... & Larson, P. (2000). Virtual environment applications in clinical neuropsychology. In K. Stanney, *The Handbook of Virtual Environments* (pp. 1027–1064). New York: Erlbaum Publishing.

Rizzo, A. S., & Kim, G. J. (2005). A SWOT analysis of the field of virtual reality rehabilitation and therapy. *Presence: Teleoperators & Virtual Environments*, 14(2), 119-146.

Rodgers, M. (1988). Dynamic biomechanics of the normal foot and ankle during walking and running. *Physical Therapy*, *68* (12), 1822–1830.

Roggendorf, J., Chen, S., Baudrexel, S., van de Loo, S., Seifried, C., & Hilker, R. (2012). Arm swing asymmetry in Parkinson's disease measured with ultrasound based motion analysis during treadmill gait. *Gait & Posture*, *35 (1)*, 116–120.

Rose, D. J., Lucchese, N., & Wiersma, L. D. (2006). Development of a multidimensional balance scale for use with functionally independent older adults. *Archives of Physical Medicine and Rehabilitation*, 87 (11), 1478-1485.

Sabatini, A. M., Martelloni, C., Scapellato, S., & Cavallo, F. (2005). Assessment of walking features from foot inertial sensing. *IEEE Transactions on Biomedical Engineering*, *52* (3), 486–494.

Salarian, A., Russmann, H., Vingerhoets, F. J., Dehollain, C., Blanc, Y., Burkhard, P. R., & Aminian, K. (2004). Gait assessment in Parkinson's disease: toward an ambulatory system for long-term monitoring. *IEEE Transactions on Biomedical Engineering*, *51* (8), 1434-1443.

Salarian, A., Russmann, H., Vingerhoets, F. J., Burkhard, P. R., & Aminian, K. (2007). Ambulatory monitoring of physical activities in patients with Parkinson's disease. *IEEE Transactions on Biomedical Engineering*, *54* (12), 2296-2299.

Satyavolu, S., Bruder, G., Willemsen, P., & Steinicke, F. (2012). Analysis of IR-based virtual reality tracking using multiple Kinects. *IEEE Virtual Reality Short Papers and Posters (VRW)*, (pp. 149-150).

Schepers, H. M., Koopman, H. F. J. M., & Veltink, P. H. (2007). Ambulatory assessment of ankle and foot dynamics. *IEEE Transactions on Biomedical Engineering*, *54* (5), 895–902.

Schmidt, R. A., & Wrisberg, C. A. (2008). Motor learning and performance: a situationbased learning approach. Human Kinetics Publishers.

Schmidt, R. L. (2005). *Motor Control and Learning: A Behavioral Emphasis*. Champaign, IL: Human Kinetics.

Schrag, A., Jahanshahi, M., & Quinn, N. (2001). How does Parkinson's disease affect quality of life? A comparison with quality of life in the general population. *Movement Disorders*, *15* (6), 1112-1118.

Schuemie, M. J., Van Der Straaten, P., Krijn, M., & Van Der Mast, C. A. (2001). Research on presence in virtual reality: a survey. *Cyberpsychology, behavior and social networking*, *4*, 183–201.

Schultheis, M. T., Rizzo, A. (2001). The application of virtual reality technology in rehabilitation. *Rehabilitation Psychology*, *46*, 296–311.

Schwenk, M., Zieschang, T., Oster, P., & Hauer, K. (2010). Dual-task performances can be improved in patients with dementia A randomized controlled trial. *Neurology*, *74* (24), 1961-1968.

Schwickert, L., Klenk, J., Stähler, A., Becker, C., & Lindemann, U. (2011). Robotic-assisted rehabilitation of proximal humerus fractures in virtual environments: a pilot study. *Zeitschrift für Gerontologie und Geriatrie*, *44* (6), 387–392.

Sen, Z., Leo, K., Tan, B., & and Tham, R. (2011). A Hybrid Human Motion Tracking System for Virtual Rehabilitation. *6th IEEE Conference on Industrial Electronics and Applications (ICIEA)*, (pp. 1993-1998).

Sergi, F., Accoto, D., Campolo, D., & Guglielmelli, E. (2008, October). Forearm orientation guidance with a vibrotactile feedback bracelet: On the directionality of tactile motor communication. *IEEE Int. Conf. Biomed. Robotics Biomechatron (BIOROB'08)*,, (pp. 433–438). Washington, DC.

Sheridan, P. L., Solomont, J., Kowall, N., & Hausdorff, J. M. (2003). Influence of executive function on locomotor function: divided attention increases gait variability in Alzheimer's disease. *Journal of the American Geriatrics Society*, *51* (11), 1633-1637.

Shumway-Cook, Woollacott, M.H. (2007). *Motor Control: Translating Research into Clinical Practice* (3rd ed.). Philadelphia, PA: Lippincott Williams & Wilkins.

Silsupadol, P., Shumway-Cook, A., Lugade, V., van Donkelaar, P., Chou, L. S., Mayr, U., & Woollacott, M. H. (2009). Effects of single-task versus dual-task training on balance performance in older adults: a double-blind, randomized controlled trial. *Archives of Physical Medicine and Rehabilitation*, *90*, 381-387.

Skog, I., Handel, P., Nilsson, J. O., & Rantakokko, J. (2010). Zero-velocity detection—an algorithm evaluation. *IEEE Transactions on Biomedical Engineering*, *57* (11), 2657–266.

Slater, M. (1999). Measuring presence: a response to the Witmer and Singer questionnaire. *Presence Teleoperators & amp Virtual Environments*, *8*, 560–566.

Snook, E. M., Motl, R. W., & Gliottoni, R. C. (2009). The effect of walking mobility on the measurement of physical activity using accelerometry in multiple sclerosis. *Clin Rehabil*, *23*, 248-258.

Sosnoff, J. J., Boes, M. K., Sandroff, B. M., Socie, M. J., Pula, J. H., & Motl, R. W. (2011). Walking and thinking in persons with multiple sclerosis who vary in disability. *Archives of physical medicine and rehabilitation*, *92* (12), 2028-2033.

Springer, S., Giladi, N., Peretz, C., Yogev, G., Simon, E. S., & Hausdorff, J. M. (2006). Dual-tasking effects on gait variability: the role of aging, falls, and executive function. *Movement Disorders*, *21* (7), 950–957.

Stanney, K. M., Mourant, R. R., & Kennedy, R. S. (1998). Human factors issues in virtual environments: a review of the literature. *Presence*, *7*, 327–351.

Stanton, R., Ada, L., Dean, C. M., & Preston, E. (2011). Biofeedback improves activities of the lower limb after stroke: a systematic review. *Journal of Physiotherapy*, *57*, 145–155.

Stone, E. E., & Skubic, M. (2011). Evaluation of an inexpensive depth camera for passive inhome fall risk assessment. *PervasiveHealth*, (pp. 71–77).

Stucki, G. E. (2002). Value and application of the ICF in rehabilitation medicine. *Disability and Rehabilitation*, *24*, 932–938.

Sullivan, K. J., Knowlton, B. J., & Dobkin, B. H. (2002). Step training with body weight support: effect of treadmill speed and practice paradigms on poststroke locomotor recovery. *Archives of Physical Medicine and Rehabilitation*, *83*, 683-691.

Suma, E. A., Lange, B., Rizzo, A., Krum, D. M., & Bolas, M. (2011). FAAST: the flexible action and articulated skeleton toolkit. *IEEE Virtual Reality Conference*, (pp. 247–248).

Sveistrup, H., McComas, J., Thornton, M., Marshall, S., Finestone, H., McCormick, A., ... & Mayhew, A. (2003). Experimental studies of virtual reality- delivered compared to conventional exercise pro- grams for rehabilitation. *CyberPsychology & Behavior*, *6*, 245–249.

Swingler, R. J., & Compston, D. A. S. (1992). The morbidity of multiple sclerosis. *The Quarterly journal of medicine*, 83 (1), 325-337.

Swinnen, E., Beckwée, D., Pinte, D., Meeusen, R., Baeyens, J. P., & Kerckhofs, E. (2012). Treadmill Training in Multiple Sclerosis: Can Body Weight Support or Robot Assistance Provide Added Value? A Systematic Review. *Multiple Sclerosis International*.

Talbot, M. L., & Junkala, J. (1981). The effects of auditorally augmented feedback on the eye-hand coordination of students with cerebral palsy. *American Journal of Occupational Therapy*, *35*, 525–528.

Tao, W., Liu, T., Zheng, R., & Feng, H. (2012). Gait analysis using wearable sensors. Sensors, 12 (2), 2255-2283.

Taylor, A. E., Saint-Cyr, J. A., & Lang, A. E. (1986). Frontal lobe dysfunction in Parkinson's disease the cortical focus of neostriatal outflow. *109* (5), 845-883.

Teng, X. F., Zhang, Y. T., Poon, C. C., & Bonato, P. (2008). Wearable medical systems for p-Health. *IEEE Reviews in Biomedical Engineering*, *1*, 62-74.

Thong, Y. K., Woolfson, M. S., Crowe, J. A., Hayes-Gill, B. R., & Jones, D. A. (2004). Numerical double integration of acceleration measurements in noise. *Measurement*, *36* (1), 73–92.

Thoumie, P., Lamotte, D., Cantalloube, S., Faucher, M., & Amarenco, G. (2005). Motor determinants of gait in 100 ambulatory patients with multiple sclerosis. *Multiple sclerosis*, *11*, 485-491.

Tombaugh, T. (2004). Trail Making Test A and B: Normative data stratified by age and education. *Archives of Clinical Neuropsychology*, *19*, 203–214.

Tong, K., & Granat, M. H. (1999). A practical gait analysis system using gyroscopes. *Medical Engineering & Physics, 21 (2),* 87–94.

Toulotte, C., Thevenon, A., Watelain, E., & Fabre, C. (2006). Identification of healthy elderly fallers and non-fallers by gait analysis under dual-task conditions. *Clinical Rehabilitation*, 20 (3), 269–276.

Trapp, B. D., Ransohoff, R. M., Fisher, E., & Rudick, R. A. (1999). Neurodegeneration in multiple sclerosis: relationship to neurological disability. The Neuroscientist, 5(1), 48-57.

Tuncel, O., Altun, K., & Barshan, B. (2009). Classifying human leg motions with uniaxial piezoelectric gyroscopes. *Sensors*, *9*, 8508–8546.

Tyler, M., Danilov, Y., Bach-y-Rita, P. (2003). Closing an open-loop control system: vestibular substitution through the tongue. *Journal of Integrative Neuroscience*, *2*, 159–164.

Valobra, G. N., Gatto, R., & Rovere, G. (2000). Medicina Manuale. Trattato di Medicina Fisica e Riabilitazione. UTET, Torino.

Van Vliet, P. M., & Wulf, G. (2006). Extrinsic feedback for motor learning after stroke: What is the evidence? *Disability and Rehabilitation*, *28*, (13–14,), 831–840.

Veltink, P. H., Slycke, P., Hemssems, J., Buschman, R., Bultstra, G., & Hermens, H. (2003). Three dimensional inertial sensing of foot movements for automatic tuning of a two-channel implantable drop-foot stimulator. *Medical Engineerng & Physics*, *25* (1), 21–28.

Verghese, J., Buschke, H., Viola, L., Katz, M., Hall, C., Kuslansky, G., & Lipton, R. (2002). Validity of divided attention tasks in predicting falls in older individuals: a preliminary study. *Journalof American Geriatric Society*, *50*, 1572–1576.

Verhoeff, L. L., Horlings, C. G., Janssen, L. J., Bridenbaugh, S. A., & Allum, J. H. (2009). Effects of biofeedback on trunk sway during dual tasking in the healthy young and elderly. *Gait Posture*, *30*, 76-81.

Verschueren, S. M. P., Swinnen, S. P., Dom, R., & De Weerdt, W. (1997). Interlimb coordination in patients with Parkinson's disease: motor learning deficits and the importance of augmented information feedback. *Experimental Brain Research*, *113* (3), 497-508.

Vuillerme, N., Chenu, O., Demongeot, J., & Payan, Y. (2007). Controlling posture using a plantar pressure-based, tongue-placed tactile biofeedback system. *Experimental Brain Research*, *179*, 409-414.

Wade, E., Parnandi, A. R., & Mataric, M. J. (2010, March). Automated administration of the wolf motor function test for post-stroke assessment. *4th International Conference on Pervasive Computing Technologies for Healthcare*.

Walker, M. L., Ringleb, S. I., Maihafer, G. C., Walker, R., Crouch, J. R., Van Lunen, B., & Morrison, S. (2010). Virtual reality-enhanced partial body weight-supported treadmill training poststroke: feasibility and effectiveness in 6 subjects. *Archives of Physical Medicine And Rehabilitation*, *91* (1), 115-122.

Wall III, C., Weinberg, M. S., Schmidt, P. B., & Krebs, D. E. (2001). Balance prosthesis based on micromechanical sensors using vibrotactile feedback of tilt. *IEEE Transactions on Biomedical Engineering*, *48 (10)*, 1153-1161.

Wall, C., & Kentala, E. (2005). Control of sway using vibrotactile feedback of body tilt in patients with moderate and severe postural control deficits. *Journal of Vestibular Research*, *15*, 313-325.

Warade, S. A., Jagannath, A., Petitpierre, C., and Sandeep U. (2012). Real-Time Detection and Tracking with Kinect. *International Conference on Computer and Information Technology*, (pp. 86-89). Bangkok.

Weikert, M., Motl, R. W., Suh, Y., McAuley, E., & Wynn, D. (2010). Accelerometry in persons with multiple sclerosis: measurement of physical activity or walking mobility? *Journal of the neurological sciences*, 290 (1), 6-11.

Weiss, P. L., Kizony, R., Feintuch, U., & Katz, N. Virtual reality in neurorehabilitation. In M. C. Selzer, *Textbook of neural repair and neurorehabilitation*. New York: Cambridge University Press.

Weiss, P. L. and Jessel, A.S. (1998). Virtual reality applications to work. Work, 11, 277–293.

Wiles, C. M., Newcombe, R. G., Fuller, K. J., Shaw, S., Furnival-Doran, J., Pickersgill, T. P.,
& Morgan, A. (2001). Controlled randomised crossover trial of the effects of physiotherapy on mobility in chronic multiple sclerosis. *Journal of Neurology, Neurosurgery & Psychiatry*, 70 (2), 174-179.

Williams, G., Doughty, K., Cameron, K., & Bradley, D. A. (1998). A smart fall and activity monitor for telecare applications. *20th Annual International Conference of the IEEE Engineering in Medicine and Biology Society*, (pp. 1151-1154). Hong Kong.

Winstein, C. (1991). Knowledge of results and motor learning: implications for physical therapy. *Journal of Physical Therapy*, *71*, 140–149.

Winter, D. A. (1984). Kinematic and kinetic pattern in human gait: variability and compensating effects. *Human Movement Science*, *3*, 51–76.

Wong, A., Lee, M., Kuo, J., & Tang, F. (1997). The development and clin- ical evaluation of a standing biofeedback trainer. *Journal of Rehabilitation Research and Development*, *34*, 322-327.

Wood, B. H., Bilclough, J. A., Bowron, A., & Walker, R. W. (2002). Incidence and prediction off alls in Parkinson's disease: a prospective multidisciplinary study. *Journal of Neurology, Neurosurgery & Psychiatry*, 72 (6), 721–725.
Woodman, O. J. (2007). *An Introduction to Inertial Navigation*. University of Cambridge. Cambridge: In-House Technical Report 696.

Wooldridge, C. P., & Russell, G. (1976). Head position training with the cerebral palsied child: an application of biofeedback techniques. *Archives Of Physical Medicine And Rehabilitation*, 57 (9), 407-414.

Woollacott, M., & Shumway-Cook, A. (2002). Attention and the control of posture and gait: a review of an emerging area of research. *Gait & Posture*, *16*, 1–14.

World Health Organization. (1981). *Disability prevention and rehabilitation*. Geneva: World Health Organization.

World Health Organization. (2001). *International Classification of Functionining, Disability and Health (ICF)*. Geneva: World Health Organization.

World Health Organization. (2007, 02 26). *http://www.who.int/features/qa/55/en/index.html*. Retrieved 12 02, 2012.

World Health Organization. (2007). Neurological Disorders: public health challenge. Brussels/Genev.

World Health Organization. (2008). *Atlas Multiple Sclerosis Resources in the World*. Geneva: World Health Organization.

Wu, G. (1997). Real-time feedback of body center of gravity for postural training of elderly patients with peripheral neuropathy. *Rehabilitation Engineering, IEEE Transactions on, 5* (4), 399-402.

Wu, G., & Xue, S. (2008). Portable preimpact fall detector with inertial sensors. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, *16* (2), 178-183.

Wu, T., & Hallett, M. Dual Task Interference in Parkinson's Diseased. *European* Neurological Review, 4 (2), 36-39.

Yamada, M., Aoyama, T., Arai, H., Nagai, K., Tanaka, B., Uemura, K., ... & Ichihashi, N. (2011). Dual-task walk is a reliable predictor of falls in robust elderly adults. *Journal of the American Geriatrics Society*, *59* (1), 163-164.

Yang, C.-C. H.-L. (2010). A Review of Accelerometry-based Wearable Motion Detectors for Physical Activity Monitoring. *Sensors*, *10* (8), 7772–7788.

Yang, Y. R., Tsai, M. P., Chuang, T. Y., Sung, W. H., & Wang, R. Y. (2008). Virtual Reality-based Training Improves Community Ambulation in Individuals with Stroke: a Randomized Controlled Trial. *Gait & Posture*, *28* (2), 201–206.

Yang, Y. R., Wang, R. Y., Chen, Y. C., & Kao, M. J. (2007). Dual-task exercise improves walking ability in chronic stroke: a randomized controlled trial. *Archives of physical medicine and rehabilitation*, 88 (10), 1236-1240.

Yogev-Seligmann, G., Giladi, N., Brozgol, M., & Hausdorff, J. M. (2012). A training program to improve gait while dual tasking in patients with parkinson's disease: a pilot study. *Archives of physical medicine and rehabilitation*, *93* (1), 176-181.

Yogev - Seligmann, G., Hausdorff, J. M., & Giladi, N. (2008). The role of executive function and attention in gait. *Movement Disorders*, *23* (3), 329-342.

Yogev, G., Giladi, N., Peretz, C., Springer, S., Simon, E. S., & Hausdorff, J. M. (2005). Dual tasking, gait rhythmicity, and Parkinson's disease: which aspects of gait are attention demanding? *European Journal of Neuroscience*, *22* (5), 1248–1256.

Yogev, G., Plotnik, M., Peretz, C., Giladi, N., & Hausdorff, J. M. (2007). Gait asymmetry in patients with Parkinson's disease and elderly fallers: when does the bilateral coordination of gait require attention? *Experimental brain research*, *177* (3), 336-346.

You, J. H., Shetty, A., Jones, T., Shields, K., Belay, Y., & Brown, D. (2009). Effects of dualtask cognitive-gait intervention on memory and gait dynamics in older adults with a history of falls: a preliminary investigation. *NeuroRehabilitation*, *24* (2), 193-198.

You, S., Neumann, U., & Azuma, R. (1999). Orientation tracking for outdoor augmented reality registration. *IEEE Computer Graphics and Applications*, 36-41.

Yun, X., Bachmann, E. R., Moore, H., & Calusdian, J. (2007). Self-contained position tracking of human movement using small inertial/magnetic sensor modules. *IEEE International Conference on Robotics and Automation*, (pp. 2526–2533.).

Zampieri, C., Salarian, A., Carlson-Kuhta, P., Aminian, K., Nutt, J. G., & Horak, F. B. (2010). The instrumented timed up and go test: potential outcome measure for disease modifying therapies in Parkinson's disease. *Journal of Neurology, Neurosurgery & Psychiatry*, *81* (2), 171–176.

Zhang, L., Sturm, J., Cremers, D., & Lee, D. Real-time Human Motion Tracking Using Multiple Depth Cameras. *IEEE/RSJ International Conference on Intelligent Robots and Systems* (pp. 2389–2395). IEEE.

Zhou, Stone, T., Hu, H., Harris, N. (2008). Use of Multiple Wearable Inertial Sensors in Upper Limb Motion Tracking. *Medical Engineering & Physics*, *30* (1), 123–133.

Zhu, R., & Zhou, Z. (2004). A real-time articulated human motion tracking using tri-axis inertial/magnetic sensors package. *IEEE Transactions on Neural Systems and Rehabilitation Engineering*, *12* (2), 295–302.

Zijlmans, J. C. M., Poels, P. J. E., Duysens, J., Van der Straaten, J., Thien, T., Van't Hof, M. A., ... & Horstink, M. W. I. M. (1996). Quantitative gait analysis in patients with vascular parkinsonism. *Movement Disorders*, *11* (5), 501–508.

Zimmerli, L., Duschau-Wicke, A., Riener, R., Mayr, A., & Lunenburger, L. (2009). Virtual reality and gait rehabilitation Augmented feedback for the Lokomat. *International Conference on Virtual Rehabilitation 2009*, (pp. 150–153).

Zuvich, R. L., McCauley, J. L., Pericak-Vance, M. A., & Haines, J. L. (2009). Genetics and pathogenesis of multiple sclerosis. *Seminars in Immunology*, *21* (6), 328-333.